

Growth discontinuities at borders

Maxim L. Pinkovskiy¹

Published online: 15 October 2016 © Springer Science+Business Media New York 2016

Abstract I document the existence of discontinuities in short- and long-term growth rates of satellite-recorded nighttime lights per capita across national borders, with growth rates of nighttime lights increasing abruptly as one crosses a border from a slower-growing country into a faster-growing one. I show that growth discontinuities are not driven by any special set of borders, or by differences in geographic and climatic conditions on the different sides of borders. I investigate multiple explanations for growth discontinuities, including differences in the determinants of growth across borders and differences in the extent to which borders form barriers to flows of goods, capital or people. I present evidence that differences in the quality of the rule of law are consistently helpful in explaining differences in growth between two countries at their border, and conclude that national-level variables such as institutions and policies may have rapid and important effects on growth.

Keywords Economic Growth and Development · Nighttime lights data · National borders · Institutions · Public goods

JEL Classification E01 · O43 · O47 · O57

1 Introduction

It is well known that rates of economic growth vary widely between countries. However, it is less well understood how growth varies within and across countries over space. Does

Maxim L. Pinkovskiy maxim.pinkovskiy@ny.frb.org

Electronic supplementary material The online version of this article (doi:10.1007/s10887-016-9139-2) contains supplementary material, which is available to authorized users.

¹ Microeconomics Function, Research Department, Federal Reserve Bank of New York, 33 Liberty Street, New York, NY 10045, USA



E. Europe, 1992-2000

Fig. 1 White areas denote lights increase and black areas denote light decrease

economic growth rise gradually as one travels through a poorer country towards a richer one, or does it change abruptly at the border? Figure 1 shows the growth in satellite-recorded nighttime lights, an intuitive indicator of economic growth, over Ukraine and its neighbors between 1992 and 2000. Areas that gained light during this time period are represented in white, whereas areas that lost light are represented in black (gray areas were unlit in both years). It is obvious that the ex-Soviet republics in this picture—Ukraine and its southern neighbor Moldova-mostly lost nighttime lights, while the Eastern European countries outside the Soviet Union-Poland, Romania and Hungary-mostly gained nighttime lights. This is consistent with the fact that during the transition from communism in the 1990s, Ukraine and Moldova contracted much more severely than did Poland, Romania and Hungary, which had exceeded their 1992 GDP by 2000. What is striking about the picture is that growth in nighttime lights appears to be discontinuous at the borders between the ex-Soviet states and their neighbors. Not only are there very few white dots in Ukraine and Moldova, and very few black dots in the Eastern European countries, but the black dots in Ukraine extend all the way to its borders with the Eastern European countries, and are immediately replaced by white dots as soon as the border is crossed. Particularly striking is that many of these national borders are relatively new. The westernmost tip of Ukraine (Ruthenia) had been part of Ukraine for only 46 years before the transition to market institutions, and had been in a political union with Hungary, Slovakia and Romania for much of the past five hundred years¹, and yet, it experienced a decline in lights as did the rest of

¹ Ruthenia had been a part of the Kingdom of Hungary since 1526, and of the Habsburg empire (which included Hungary, Slovakia, and the parts of Poland and Romania that are visible in this picture) since 1699. After the collapse of the Habsburg empire as a result of World War I, Ruthenia became a part of Czechoslovakia in 1918. Ruthenia was annexed by the Soviet Union in 1945 as a result of World War II, and attached to the Ukrainian Socialist Soviet Republic, which became the independent country of Ukraine in 1991.

Ukraine, while the neighboring parts of Hungary, Slovakia and Romania experienced growth in lights. Moreover, Moldova and Romania share the same language, religion and culture (although they have been politically separate for most of their modern history), but have had radically different lights growth experiences in the 1990s with Romania's lights increasing, Moldova's lights declining, and the growth experience changing discontinuously at the border.

This paper presents evidence that Fig. 1 is not an anomaly, but part of a general pattern of discontinuities in the rates of economic growth around national borders. The key methodological tool that enables me to establish the presence of growth discontinuities at borders is measuring economic growth in narrow bands around national borders by computing growth rates of satellite-recorded nighttime lights in these regions. Night lights are an ideal and indispensable data source for this project because they are one of the few indicators of economic activity that exist at a sufficiently fine resolution to allow the analysis of narrow neighborhoods around national borders, as well as because the method of their collection is continuous across national borders. National accounts data is fundamentally unsuitable for such a project because it is available, at best, only at a regional level, and does not permit considering regions other than large political subdivisions, thus making it inappropriate for a regression discontinuity analysis. Survey data may overcome this problem as it samples individuals or villages rather than geographical units, but it would introduce an artificial discontinuity at borders because each survey is conducted within a single country. Hence, errors attributable to the questionnaire or to the performance of different survey teams would be different on different sides of national borders. On the other hand, even if there is important spatial heterogeneity in the way that satellites record lights data, the remote sensing process should be continuous, and therefore, roughly stable within a neighborhood around a given border.²

Using satellite data on nighttime lights as a proxy for economic activity around borders, I document a strong and statistically significant relationship between differentials in growth of lights per capita across a border and differences in growth in the bordering countries over 5- and 18-year periods between 1992 and 2010. As one moves from a slower-growing to a faster-growing country, the 5-year growth rate of lights per capita rises on average by 1.48 percentage points (or by 0.13 standard deviations) and the 18-year growth rate of lights per capita rises on average by 1.28 percentage points (or 0.19 standard deviations). For every 1 percentage point difference in the 5-year growth rates of GDP per capita of two bordering countries at their mutual border, and for 18-year growth rates, there is a 0.45 percentage point difference. Therefore, there is a sizeable elasticity (between a third and a half, though less than unity) of growth differentials at borders to growth differentials nationwide. Moreover, these differentials are not driven by borders in any particular region, suggesting that they are not results of composition bias, but rather an enduring feature of the data to be explained.

After establishing the existence of border discontinuities in economic growth around the world, I examine the factors that might explain these discontinuities. One immediate explanation for why we observe border discontinuities might be that borders are drawn endogenously, with terrain and climatic conditions that are more favorable for growth being predominantly located on the sides of borders that belong to faster-growing coun-

 $^{^2}$ The recorded brightness of lights may depend on cloud cover, humidity and other atmospheric conditions in a region, but it is implausible that national borders consistently conform to atmospheric fronts. Robustness checks with controls for temprerature, precipitation, altitude and slope on both sides of borders do not alter the results.

tries. Borders could also be drawn endogenously if they are drawn around populations with particularly high or low growth rates, such as when a high-growing region secedes from a country. In this case, the discontinuity in growth rates causes the border, rather than the other way around. I find weak evidence that discontinuities in economic growth tend to be stronger at borders with above-median rather than below-median differences in key geographic covariates, but the order of magnitude of border discontinuities is similar for borders whose sides are geographically similar as well as geographically different. Moreover, there is no systematic tendency for certain kinds of geographies to coincide with faster- rather than slower-growing sides of borders. While it is difficult to test whether borders may have endogenously been drawn to separate groups of different productivies and propensities for growth because growth data at borders is not available historically, I show that estimated growth discontinuities at borders are of similar magnitude for old and recent borders, curved and straight borders, borders over rugged or flat terrain and borders near or far away from capital cities, which indirectly suggests that the history of the formation of a border matters little for the the growth discontinuity across that border. Furthermore, since I consider differences in economic growth in narrow neighborhoods around borders, so long as the location of borders in these neighborhoods is not precisely determined with respect to economic growth, endogeneity in border placement should not be the driver of growth discontinuities at borders.

Aside from endogenous border placement, growth discontinuities at national borders is consistent with either (or both) of the two following hypotheses:

- 1. The growth process is not smooth across space, but depends on national-level shocks or national-level potential determinants of growth (such as national institutions, the national stock of human capital, culture or history) that vary discontinuously across borders. In particular, models in which local institutions or local culture are key to economic growth, but vary substantially across space and straddle borders would not generate growth discontinuities.³ The border must form a barrier to the ability of people to avail themselves of particular institutions, to cultural influences, to the ability to obtain an education (or work with people with high human capital), or to the operation of other determinants of growth.
- 2. Even in the absence of any differences in growth determinants across borders, there may still be discontinuities in rates of economic growth because borders form barriers to random economic shocks. Since different sides of a national border often have different languages, cultures and regulatory regimes, and often have explicit discouragements to trade across the border because of tariffs, the economics on the two sides of the border are less connected by trade than two economically comparable regions within the same country. Therefore, a shock in one country—such as a fall in demand for oil or a poor harvest—may fail to spill over a national border because of the discontinuous increase in transaction costs along that border.

I first show that borders that present greater barriers to economic flows—such as borders with low trade, migration and remittance flows, and high tariffs—tend to have the same

³ Michalopoulos and Papaioannou (2014) implicitly consider a model with this property when investigating the lack of a discontinuity in economic activity at African borders at which an ethnicity was split between two countries during the Scramble for Africa. The find that, at least in Africa, local institutions, as proxied by ethnic institutions, matter more than national institutions (proxied by rule of law as measured by the World Bank) do. As I look at a model of growth rates rather than levels, and as I do not analyze borders that split rather than separate ethnicities, I do not claim that my results contradict Michalopoulos and Papaioannou (2014). In Sect. 5.2 I find some evidence that cultural similarities across borders in genetic distance from the UK or in the official language are correlated with smaller growth discontinuities at borders.

149

magnitudes of border discontinuities as do borders with low barriers to economic flows. On the other hand, borders separating culturally more homogenous populations—such as those having the same language, or those that have had been colonized by the same country—tend to display smaller growth discontinuities. I then perform a correlational analysis of which of the determinants of economic growth that are most frequently cited in the literature help account for growth discontinuities at borders. I find that the rule of law measure from the World Governance Indicators (WGI) is correlated with growth of lights at the border, even conditional on various other covariates, while public goods provision, human capital and trust are not correlated with growth discontinuities conditional on institutional quality.

This paper is closest to Michalopoulos and Papaioannou (2014), who use night lights data to look at the (non)importance of national institutions for economic activity within African ethnicities split by national borders far away from their countries' capitals. The major innovation of this paper is to consider discontinuities in *growth rates*, rather than levels, of economic activity across national borders. This innovation is valuable, since finding border discontinuities in economic growth rates, which tend to be volatile and much less persistent than is the level of economic activity, suggests that if these discontinuities can be attributed to policy variables, such as institutions, then policy might have nontrivial effects on economic growth.

The paper is organized as follows: Sect. 2 describes the data. Section 3 provides baseline results on border discontinuities in growth rates, as well as robustness checks to varying the econometric procedures and accounting for basic covariates that may be different on different sides of borders. Section 4 explores the heterogeneity of growth discontinuities across borders and provides suggestive evidence that they are not driven by a special subset of borders and that they are not explained by faster-growing sides of borders having different geography from the slower-growing sides of borders. Section 5 explores whether growth discontinuities at borders may be explained by the discontinuous barriers to economic interchange that borders pose. Section 6 discusses determinants of economic growth that may help statistically explain growth discontinuities at borders. Section 7 concludes.

2 Description of the data

In this section, I discuss the data sources used in this paper that are the most key to its contribution. Other sources, whose use is more standard, are described in Online Appendix 0.

2.1 The night lights dataset

Data on light radiance at night is collected by the DMSP-OLS satellite program and is maintained and processed by the Earth Observation Group and the NOAA National Geophysical Data Center. Satellites orbit the Earth every day, recording images of every location between 65 degrees south latitude and 65 degrees north latitude at a resolution of 30 arcseconds (approximately 1 square km at the equator) at approximately 8:30-10 PM local time. The images are processed to remove cloud cover, snow and ephemeral lights (such as forest fires) to produce the final product.⁴

Each pixel (1 square kilometer) in the radiance data is assigned a digital number (DN) representing its brightness. The DNs are integers ranging from 0 to 63, with higher digital numbers corresponding to greater light density. Recent papers using nighttime lights data have constructed light density as some aggregate of powers of pixel DN's in a region: for

⁴ The nighttime lights maps are available for download from the URL http://www.ngdc.noaa.gov/dmsp/ downloadV4composites.html.

example, Henderson et al. (2011) and Michalopoulos and Papaioannou (2014) use the sum of all DNs, while Chen and Nordhaus (2010) use the sum of all DNs taken to the power 3/2. However, pixels with DN equal to 0 or 63 may be top- or bottom-censored. Another known problem with the lights data is the presence of overglow and blooming: light tends to travel to pixels outside of those in which it originates, and light tends to be magnified over certain terrain types such as water and snow cover. Overglow and blooming tend to make nearby pixels more similarly lit than they should be, thus working against finding discontinuities across borders.

The night lights dataset has been extensively analyzed in the remote sensing literature for its utility in predicting economic activity; see Elvidge et al. (1997), Sutton et al. (2007), Doll (2006), Ghosh et al. (2010) and Elvidge et al. (2012). Baugh et al. (2009) thoroughly describes the construction of the night lights dataset, and Doll (2008) comprehensively discusses its uses and pitfalls. Its pioneering use in the economics literature has been Henderson et al. (2011). Chen and Nordhaus (2010) discuss the limitations of the lights dataset; in particular, they argue that the relationship between light density and output density becomes uninformative because of top-censoring and bottom-censoring at DN = 63 and DN = 0. Pinkovskiy and Sala-i-Martin (2016) use nighttime lights as an independent benchmark to assess the quality of household survey means relative to national accounts data, and find national accounts to be more reliable than household surveys. Michalopoulos and Papaioannou (2014) use the night lights dataset to construct a proxy for output per capita in African ethnic territories to assess the consequences of partitioning ethnicities during the Scramble for Africa, and Alesina et al. (2016) use them to construct a proxy for incomes of ethnic groups more generally. Jerven (2013) points out that national income statistics may be notoriously mismeasured in parts of the developing world (e.g. Africa), suggesting that nighttime lights might even be more accurate than official economic statistics for such places.

2.2 Gridded population of the world data

The Gridded Population of the World (GPW) dataset is constructed and maintained by the Socioeconomic Data and Applications Center (SEDAC) at the Center for International Earth Science Information Network at the Earth Institute at Columbia University. The dataset compiles population information from national censuses for political units (typically very small ones, like municipalities and census tracts, but in some cases, much larger ones) in order to achieve its resolution. Within a political unit, population is distributed uniformly. In Sect. 3 I demonstrate that differences in the resolution of the population data across countries do not impact my results substantially.

3 Baseline results

3.1 The dependent variable

3.1.1 Calibration

To convert lights data into a single quantity comparable to GDP, I assume a low-parameter function approximating the relationship between DN and output density, and calibrate its parameters using aggregate light density for countries and national accounts data on GDP per capita in 2005 international dollars. Specifically, I estimate the parameters of the function using nonlinear least squares, in which I try to explain GDP density per unit area in a country

with measures of light density for the country constructed using pixel digital numbers. The assumed relationship is

$$\ln\left(1+y_{i}\right) = c + \ln\left(1+c_{0}*v_{0,i}+c_{b}*\sum_{j=0}^{63}\left(j^{d}\right)*v_{j,i}+c_{1}*v_{63,i}\right) + \varepsilon_{i} \qquad (1)$$

where *i* indexes countries, y_i is GDP density of country *i* (obtained from the World Bank), $v_{j,i}$ is the fraction of pixels with digital number equal to *j* in country *i*, and ε_i is the error term. I use the transfer function $\ln (1 + x)$ rather than $\ln (x)$ because the latter is not defined for x = 0 while the former is defined for all nonnegative *x* (and some negative values of *x* as well). This is not a problem in the estimation of the calibration equation, as the light density of no country is equal to zero; however, the light density at some borders does attain the value zero, which explains the need to use such a specification. This specification is similar in spirit to the pixel-level specifications in Michalopoulos and Papaioannou (2014), which also consider variations in light density across small regions around national or ethnic borders.

I estimate Eq. (1) for every satellite-year in the DMSP-OLS dataset, setting $c = c_0 = c_1 = 0$, $c_b = 1$ and d = 3/2 as my initial values.⁵ For multiple years (in particular 2000 and 2005), the estimates of the top-censoring and bottom-censoring coefficients c_0 and c_1 are equal to zero, suggesting that top-censoring and bottom-censoring is not a particularly important limitation of the data.⁶

3.1.2 Computation of the dependent variable

To obtain a lights-based proxy for economic growth around a given border for a given year, I construct neighborhoods containing all points whose shortest distance to the border is less than X kilometers, where X ranges from 10 to 70 in increments of 10 kilometers. Since national borders tend to differ substantially in length, I standardize them and the neighborhoods around them by dividing each border and its neighborhood into pieces corresponding to their intersections with a 1-degree by 1-degree grid superimposed on the world map of country borders and neighborhoods (CShapes, Weidemann et al. 2011). Having started with 270 borders, I obtain 1352 border pieces, each defined by the intersection of a border with

⁶ In addition to measuring the growth rate of calibrated lights per capita, I also present results in which the dependent variable is constructed as the growth in light density, which is measured as

$$\ln\left(0.01 + \sum_{j=0}^{63} j * v_{j,i}\right) \tag{2}$$

following Michalopoulos and Papaioannou (2014). The resulting dependent variable has a correlation of greater than 0.98 with the lights per capita variable, even conditional on other covariates of my baseline specification (see sect. 3.4). Therefore, I present only results for lights per capita in the main text, and relegate results for light density to Appendix I. My results typically are robust across the different ways of computing the dependent variable. The similarity of results using light density and results using lights per capita is consistent with population not being disproportionately low or high on faster growing sides of borders, which is one of the findings in this paper.

⁵ Chen and Nordhaus (2010) use d = 3/2 in their related, though different, specification of the relationship between DN and output density.

I conduct the estimation by performing nonlinear least squares using the Matlab routine **fmincon**, in which I impose the constraint that c_0 and c_1 cannot be negative.

the rectangular grid.⁷ For each border piece, there are 7 associated neighborhood pieces (corresponding to the 7 neighborhoods around each border, partitioned by the grid) and 2 associated sides of the border piece within each neighborhood (corresponding to the two sides of the border, partitioned by the grid).⁸ In all my computations of standard errors, I cluster the standard errors by border (rather than border piece) so I have 270 clusters. I then use an ArcGIS Python script to compute the fraction of pixels with each digital number for each side of the border within each neighborhood. Subsequently, I use the calibrated values of c, c_0, c_1, c_b and d to compute the right-hand side of Eq. (1) for each side of the border, thus obtaining a proxy for the output density of the given country within X kilometers of the given border. Finally, I multiply by the area and divide by the population of this region (as discussed in Sect. 2, I obtain population at nominal 2.5 arcminute resolution from the Gridded Population of the World dataset) to obtain a lights-based estimate of GDP per capita for the given country within X kilometers of the given border. I then compute the average annualized growth rate of this variable for the time intervals at which population data is available (1992-1995, 1995-2000, 2000-2005 and 2005-2010) as well as for the overall 18-year interval 1992-2010.

There are good reasons to expect that this calibration procedure creates a variable that is, on average, close to the true value of GDP growth per capita in the region of interest. Henderson et al. (2011) and Chen and Nordhaus (2010) document the tight association between GDP density and measures of light density that are similar to the one used in this paper. The fit of the selected specifications to the GDP data for countries is very good as the first panel of Fig. 4 attests: the lights explain approximately 73 % of the variation in GDP per capita, and the plot looks approximately linear. I also present the relationship between the 18-year growth of the calibrated lights series and the growth of GDP per capita in the second panel of Fig. 4. The fit is not as good (the lights explain only 23 % of the variation in growth), but still quite strong, and the positive correlation is unmistakeable.

3.2 Descriptive analysis and graphs

3.2.1 Descriptive statistics

Table 1 provides descriptive statistics for the main variables of interest: growth rates of lights per capita at borders for 5- and 18-year periods, growth of lights per capita and GDP per capita nationwide, and some covariates. I present the mean and the standard deviation of each variable, as well as the mean and the standard deviation of each variable computed on the side of each border corresponding to the faster-growing and slower-growing side of the border, respectively, and the mean and standard deviation of the difference. The descriptive statistics foreshadow more formal results. We immediately see that 5- and 18-year growth rates of lights per capita are higher on the sides of borders corresponding to faster-growing countries. The averages of these growth rates at the border are somewhat lower than the averages of these growth rates nationwide, but are approximately the same level as the averages of World

⁷ This grid does not coincide with integer lines of latitude and longitude, so there are no problems of the grid falling exactly upon a border.

⁸ It is useful to recognize that the sample of borders overrepresents landlocked countries, because the latter have more neighbors than coastal countries by definition, and completely excludes countries that are islands (like Japan or Australia). While landlocked countries tend to be poorer than coastal countries, it is unclear whether this should affect the difference-based estimates in the paper. Estimates by region tend to be similar to the global estimates. Moreover, population-weighting the observations should reduce the bias towards countries that are less developed and support lower population densities.

Table 1 Descriptive statistics

	Mean, overall	Mean, Low Ntwide side	Mean, high Ntwide side	Difference	Number of borders
Panel 1: Growth of lights on the	e border and	nationwide			
5-Year Growth Lights per	2.68	1.70	3.69	2.08	270
Capita, Border	(7.35)	(7.31)	(7.25)	(7.85)	
18-Year growth lights per capita, border	2.40	1.77	3.05	1.36	270
	(4.52)	(4.23)	(4.72)	(4.23)	
5-Year growth GDP per capita, Ntwide	2.62	1.30	3.97	2.75	270
	(3.24)	(2.98)	(2.91)	(2.68)	
18-Year growth GDP per capita, Ntwide	2.47	1.53	3.43	1.96	270
	(1.90)	(1.66)	(1.64)	(1.45)	
5-Year growth lights per capita, Ntwide	3.47	1.53	5.46	4.07	270
	(5.14)	(4.59)	(4.91)	(3.91)	
18-Year growth lights per capita, Ntwide	3.31	2.01	4.64	2.74	270
	(3.11)	(2.89)	(2.75)	(1.82)	
	Mean, Overall	Mean, Low Ntwide Growth GDP Side	Mean, High Ntwide Growth GDP Side	Difference	Number of Borders
Panel 2: Covariates					
Log total road length	3.18	3.13	3.22	.08	270
	(1.79)	(1.82)	(1.75)	(1.66)	
Log population	15.70	15.68	15.72	.04	270
	(2.12)	(2.11)	(2.14)	(2.15)	
Rule of law, WB	36	55	16	.40	266
	(.91)	(.89)	(.90)	(.88)	
Percent of roads paved, WDI	27	35	19	.17	264
	(.88)	(.96)	(.78)	(.82)	
Average years of education, BL	56	61	51	.06	186
	(.91)	(.89)	(.92)	(.59)	
Trust, WVS	52.00	49.76	53.77	6.65	159
	(23.49)	(21.33)	(24.93)	(28.52)	

Descriptive statistics. Standard deviations of all variables are in parentheses below their mean values. There are two observations per border piece: one for the slower-growing side (according to national GDP), and one for the faster-growing side. Data for all spatial variables including the growth rate of lights at borders, roads and population are for 70-km neighborhoods around the border



Fig. 3 Discontinuity plots: growth in light density across selected Eastern European borders

Bank-recorded growth rates of GDP per capita. In panel 2 of the table, where we look at institutions and public goods, we see that the rule of law of the faster-growing country at a border is much better on average than the rule of law of the slower-growing country at that border. Public goods (proxied by the fraction of roads paved) are also better on average in the faster-growing country at a border, but the difference varies considerably across borders, as it does for trust and the number of years of schooling. Interestingly, local public goods (log roads near the border) are very close to each other on the faster-growing and slower-growing sides of borders on average.

3.2.2 Elementary discontinuity plots and correlations

I present several elementary graphs that suggest large discontinuities in growth rates of economic activity at national borders. First, Fig. 3 provides a basic illustration of my methodology by plotting the growth in nighttime lights per capita as a function of distance to the border for some of the Eastern European borders in Fig. 1. The discontinuities in the growth rates at the Romania-Moldova and the Romania-Ukraine borders are unmistakeable, and capture in a graph the stark message that Fig. 1 captures in a map. Panel 1 of Fig. 5 shows a more global illustration of growth discontinuities by presenting a discontinuity plot of the 5-year growth



Fig. 4 Correlation of World Bank GDP with calibrated lights in levels and growth rates



Fig. 5 Discontinuity plots: average discontinuity in the growth of lights per capita at borders

rate of lights per capita (predicted GDP per capita using lights) against distance from border in the direction of the faster-growing country (in terms of national GDP per capita) at the border. To construct this plot, I effectively compute a population-weighted average of plots like the ones in Fig. 3. I identify the faster-growing country and the slower-growing country at each border according to which one has the higher 5-year growth rate of World Bank GDP per capita. I then average the growth rates at the border (weighting by population in the relevant neighborhood of the border) of the slower-growing countries for 10-km intervals of distance to the border. I repeat the same procedure for the faster-growing countries and plot the averages as a function of distance to the border either for the faster-growing (on the right) or for the slower-growing (on the left) side.

It is apparent that there is a discontinuity at the border crossing point, with the side of the border corresponding to the faster-growing country having a 5-year border growth rate about 1.5 percentage points higher than the side corresponding to the slower-growing country. The last point on the poorer side (at -10 km) is approaching the points on the richer side, but the other points on the poorer side are far removed from those on the richer side (by at least the 1.5 percentage points of the discontinuity). This can be explained by overglow in the data: light from the faster-growing side of the border illuminates the slower-growing side, making it appear to be growing faster than it really is. Panel 2 presents a similar discontinuity plot for the 18-year growth rate. Both plots are qualitatively the same: the growth rate of lights is substantially smaller for the sides of borders corresponding to the slower-growing countries,



Fig. 6 Correlation plots, growth of lights per capita at borders



Fig. 7 Discontinuity plots, growth of lights per capita at selected borders

with some convergence at the last 10 km, which is likely because of overglow. The growth discontinuities appear to be on the order of 1 percentage point or larger.

It is worth considering a few examples of growth discontinuities at borders outside Eastern Europe. Panel 1 of Fig. 7 shows the discontinuity at the border between Zimbabwe and Botswana in Africa. Botswana is one of the few middle-income countries in Africa and enjoys relatively rapid growth of about 2 % per year, while Zimbabwe's economy has shrunk at a similar rate between 1992 and 2010 owing to the expropriations and hyperinflationary

policy carried out by Robert Mugabe. We see that Botswana's growth in nighttime lights at the border is much higher than Zimbabwe's. (The fact that Botswana's growth falls as one moves away from the border while Zimbabwe's rises is likely explained by Zimbabwe's population rising rapidly about 60 kilometers away from the border, while Botswana's population falling off at a similar distance). Panel 2 shows the border between Cambodia and Vietnam. During the time period under consideration, Vietnam had implemented market reforms and increased trade with other Southeast Asian countries as well as with the United States, while Cambodia has struggled with political instability and with the legacy of a brutal dictatorship in the 1970s. We see that nighttime lights grow discretely faster in Vietnam than in Cambodia at their joint border. Panel 3 shows the border between Colombia and Venezuela. Over the period 1992 to 2010, the Colombian government was largely successful in ending a longrunning civil war and asserting control of important parts of the country, while Venezuela experienced severe economic mismanagement under the regime of Hugo Chavez. Likely as a result of these trends, the Colombian economy grew at about 1.5 % per year, while the Venezuelan economy shrank on average. We see that at the border, growth in nighttime lights in Colombia is stably higher than it is in Venezuela. The magnitude of this discontinuity is around 2 percentage points in average annualized growth, which is close to the average magnitude of a growth discontinuity at a border. Lastly, Panel 4 shows the border between the U.S. and Mexico. Here, although the U.S. grew faster than Mexico during this period (1.4 % per year vs. 0.9 % per year), there appears to be no discontinuity in growth rates of nighttime lights at the border.⁹ This is likely because northern Mexico receives large amounts of foreign investment from the U.S., and has been growing at a much faster rate than Mexico as a whole. In Sect. 5 we will consider the possibility that cross-border economic flows or their absence account for the absence or presence of growth discontinuities at borders, and we will conclude that they do not account for much variation in the magnitude of border discontinuities. Hence, the case of the U.S.-Mexico border is likely an exceptional one.

The discontinuity plots in Fig. 5 elide the fact that countries are extremely heterogeneous and the difference in economic activity at the border between a faster-growing and a slowergrowing country may vary widely, even if it is large and positive on average. One would instead expect the difference at the border to be somehow related to the difference in economic growth between the two countries overall: countries with wide disparities in growth rates (like Ukraine and Romania) should have larger discontinuities at their common border than countries with similar growth rates (like France and Germany). Figure 6 presents plots of differences in growth rates in lights per capita at borders¹⁰ against differences in growth rates of GDP per capita in the bordering countries (each border difference being weighted by total population at the border, and with outliers as defined in Table 3 removed). The positive correlations and their strength are apparent, even though the dispersion of the data is large.

3.3 Baseline results

I now present formal analysis to document economic discontinuities at borders. I first run regressions at each border piece in each year of the form

$$g_{i,b,t,d} = \tilde{g}_{i,b,t} + \delta_{i,b,t}d + \eta_{i,b,t,d}, \text{ weighted by } 1 \left(d \le h_{i,b,t} \right)$$
(3)

⁹ If we were to look at the levels, rather than the growth rates of nighttime lights, we would see lights in Mexico at the border be much lower than lights in the U.S. at the border, except for the last 10 miles of the Mexican border, where the lights would be equal to the U.S. lights. These last 10 miles are likely accounted for by overglow, border-specific infrastructure, or an especially high intensity of foreign investment even relative to the rest of northern Mexico.

¹⁰ These are local linear discontinuity estimates, whose construction is discussed below.

where $g_{i,b,t,d}$ is the value of the growth of lights per capita¹¹ in country *i*, year *t*, border piece *b* and at distance *d* away from the border in border piece *b*. The parameters to be estimated are $\tilde{g}_{i,b,t}$, the value of *g* at the border in country *i*, and $\tilde{\delta}_{i,b,t}$, the slope of *g* at the border. I choose the bandwidth $h_{i,b,t}$ to assign greater weight to observations close to the border using a cross-validation procedure that selects the bandwidth to maximize the accuracy of predictions 10 km away from the border for each regression I run. I therefore fit a separate local linear regression and choose a separate bandwidth in each country-border piece-year.

A complication in using local polynomial estimation to calculate border discontinuities with nighttime lights data is that the data generating process does not obey the standard assumption of independently generated data with the number of observations at each site going to infinity. Instead, the nighttime lights data constitute a global census of visible nighttime lights, taken at a fixed resolution. Therefore, the asymptotics for the regression discontinuity estimator must be calculated as the resolution of the data goes to infinity (the pixel size goes to zero) rather than as the domain of the pixels expands to infinity. Such an asymptotic scheme is referred to as *infill asymptotics* in the spatial econometrics literature. A natural assumption for such data is that the errors from trend of neighboring data points are correlated. A contribution of this paper is to derive the properties of the local polynomial estimator under infill asymptotics with correlated errors. I prove that with very general assumptions on the covariance structure of the outcome variable the local polynomial estimator is consistent, and has a smaller asymptotic variance than it would if the errors from trend were independent¹². Intuitively, the local polynomial estimator exploits the correlation in the errors, so that only their unpredictable component contributes to the asymptotic variance. In the special case that the error from trend is mean-square continuous (has no unpredictable component), the local polynomial estimator converges at a nonstandard rate of $1/\sqrt{h}$, where h is its bandwidth. I describe this estimator in detail in Online Appendix II.

Having obtained estimates of growth in lights and of its slope at borders, I run regressions of the form

$$\tilde{g}_{i,b,t} = \alpha_{b,t} + \beta u_{i,b,t} + \varepsilon_{i,b,t} \tag{4}$$

$$\tilde{g}_{i,b,t} = \chi_{b,t} + \gamma g_{i,t} + \upsilon_{i,b,t}$$
(5)

where $\tilde{g}_{i,b,t}$ is the local linear estimate of the growth rate of lights per capita at border piece *b* in country *i* and year *t*, $g_{i,t}$ is the corresponding growth rate of GDP per capita for country *i* as a whole, $u_{i,b,t}$ is an indicator that country *i* has the larger corresponding growth rate of GDP per capita of the two countries at border piece *b*, $\alpha_{b,t}$ and $\chi_{b,t}$ are the border piece-year fixed effects in each regression, and $\varepsilon_{i,b,t}$ and $\upsilon_{i,b,t}$ are the error terms in each regression. The parameters of interest to be estimated are β , the average percentage point rise in the growth rate of economic activity as one crosses a border from a slower-growing to a faster-growing country, and γ , the elasticity of the difference of the growth rate of output at the border to the difference of the growth rate of GDP per capita of the border in a 70-km neighborhood on both sides of the border (so nearly empty borders, like those through the Sahara desert or in the Amazon rainforest, get very little weight), and I cluster all standard errors by border (not border piece). Finally, in order to account for the error of the first-step estimation of the growth rates of light at borders, I augment the variance of the regression by the first-step variances of the dependent variables from the local linear estimation as described in Online Appendix II.

¹¹ I perform all analyses separately for 5-year growth rates and 18-year growth rates.

¹² This result is most closely related to Card and Lee (2008).

It is worth noting that one can take the difference the two observations within a border piece b at time t in order to get rid of the fixed effects in the second-step regressions. Such a procedure would yield the following regressions:

$$\Delta \tilde{g}_{i,b,t} = \beta \Delta u_{i,b,t} + \Delta \varepsilon_{i,b,t} \tag{6}$$

$$\Delta \tilde{g}_{i,b,t} = \gamma \Delta g_{i,t} + \Delta \upsilon_{i,b,t} \tag{7}$$

These simple linear regressions produce numerically identical estimates of β and γ to the fixed-effects regressions (4) and (5), but are much easier to interpret: in the first case, we regress the difference in growth of lights at the border between the side corresponding to the faster-growing country and the side corresponding to the slower-growing country on a constant, and in the second case, we regress the difference in the growth of lights across the border on the difference in the growth of GDP per capita between the bordering countries without a constant. Hence, the parameters β and γ are, respectively, the average percentage point rise in the growth rate of economic activity as one crosses a border from a slower-growing to a faster-growing country, and the fraction of the difference of the growth rate of GDP per capita of the bordering nations (in percentage terms) that persists to the border.¹³

Table 2 presents estimates of β and γ for 5-year and 18-year growth rates. The first row shows the baseline estimates. We see that average annualized 5-year growth in calibrated lights per capita jumps on average by 1.48 percentage points upon crossing from a slowergrowing into a faster growing country. We also see that a 1 percentage point difference in national GDP per capita average annualized 5-year growth rates translates into a 0.38 percentage point difference in calibrated lights per capita growth rates at the countries' joint border. The estimates when 18-year growth rates are used remain similar: β is 1.28 and γ is 0.45. All estimates are significant at least at the 5 % significance level. We can assess the magnitudes of these estimates from the second row of the table, which standardizes growth at the border and nationwide growth to have unit variance. We see that growth as measured by nighttime lights jumps by 0.13 to 0.19 standard deviations at the border between a slower-growing country and a faster-growing country, while a 1-standard deviation in the difference between national growth rates of two bordering countries is associated with about a 0.11 to 0.13 standard deviation jump in the difference between growth rates at the border. These numbers are likely to be lower bounds because the lights proxies very likely measure economic growth at borders with a large amount of noise; the standard deviation of growth rates at the border is nearly four times the size of the standard deviation of nationwide growth rates. The average bandwidth used to obtain these estimates is large, about 49 km when 18-year growth of lights per capita is used as the dependent variable. (I restrict my analysis to a 70-km neighborhood of the border, and allow a maximum bandwidth size of 60 km, which is attained for slightly under half of my observations). To assess the sensitivity of my estimates to bandwidth choice, I provide results with a bandwidth of 30 km for all countries. No coefficients lose significance, with the coefficients on 5-year growth rates and 18-year growth rates of lights per capita declining slightly in magnitude and the coefficients on the 18-year growth rates of light density rising slightly in magnitude.

Having observed growth discontinuities at borders, a natural falsification exercise is to re-run my regressions using fake borders, for which I should not expect a discontinuity. I perform two such exercises: one in which I draw the fake borders at a distance of 30 km from the real borders into the interior of the faster-growing country, and one in which I draw the fake borders at a distance of 30 km into the interior of the slower-growing country.

¹³ In particular, it is easy to see that if growth were continuous at borders, then γ would be zero.

Dep. Var. Ind. Var.	(1)5-Year growth lights p/cIndicator	(2) 5-Year growth lights p/c Elasticity	(3) 18-Year growth lights p/c Indicator	(4) 18-Year growth lights p/c Elasticity
Baseline estimates (H=51)	1.48***	.38***	1.28**	.45**
	(.51)	(.11)	(.57)	(.20)
Beta coefficients	.13***	.11***	.19**	.13**
	(.04)	(.03)	(.08)	(.06)
Estimates with BW=30 km	1.13**	.26**	1.18***	.43**
	(.52)	(.10)	(.44)	(.17)
Placebo estimates at -30 km	92	20	60	08
	(.60)	(.13)	(.57)	(.22)
Placebo estimates at +30 km	69	21**	48	19
	(.46)	(.09)	(.44)	(.15)
FE Estimates in 50 km Buffer	1.97***	.44***	1.61***	.57***
	(.51)	(.12)	(.54)	(.19)
Border-year fixed effects	Yes	Yes	Yes	Yes
No. observations	5408	5408	1352	1352
No. borders	270	270	270	270
R^2 of first row	.01	.01	.05	.04

Table 2 Local linear estimates of growth discontinuities at borders

Data on lights and population available from the NOAA and CIESIN, respectively, for 1990 (1992 for lights), 1995, 2000, 2005 and 2010. Each column presents two-step estimates of Eq. (4) or (5). Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account two-step estimation and infill asymptotics in parentheses. Each observation weighted by total population in the respective buffer piece. The dependent variable in each specification is the Eq. (3) local linear estimate of the indicated concept of growth of lights at the border corresponding to the given country-border piece year. The independent variable, whose coefficient is given in each cell, is the indicator that a given side of the border corresponds to the country with the higher corresponding national growth rate (5-year or 18-year) of GDP per capita from WDI, or the corresponding national growth rate itself. The average bandwidth for the local linear regressions in row 1 is 49. Rows 4 and 5 compute discontinuities at fake borders that are 30 km into the slower-growing country, respectively, away from the true border. The fixed-effect specification is a one-step fixed-effects regression with growth of lights per capita or growth of light density in a 50-km neighborhood of the border as the dependent variable

The discontinuity estimates for the fake borders are statistically insignificant (except in one case out of 8), which is reassuring, but the magnitudes of some of these estimates are large (half or more of the baseline estimates), which is disconcerting. However, all the large, or statistically significant, estimates are negative, suggesting that growth rates *fall* as one approaches closer than 30 km to the border from the slower-growing side, or goes further than 30 km away from the border into the faster-growing side, which is not consistent with growth rising continuously as one moves closer to or further into the faster-growing side of the border. Finally, Row 6 presents estimates in which no local linear regression is used, and the dependent variables $\tilde{g}_{i,b,t}$ are defined to be growth rates of lights per capita in a 50-km neighborhood of border *b* on the side of country *i* in year *t*. This methodology is more conventional than the two-step procedure that I employ for my baseline estimates, and is also more likely to be robust to overglow, but it may fail to capture convergence of growth rates at the border and find a discontinuity where none exists. With this methodology,

we obtain discontinuity estimates that are very similar to the baseline estimates for 18-year growth rates and are about 25 % larger than the baseline estimates for the 5-year growth rates, which suggests that if the local linear procedure captures any convergence of light growth at the border (either because of overglow or because economic interchange smooths out discontinuities), this convergence is not particularly important.

3.4 Elementary robustness checks

Table 3 presents robustness checks of the baseline discontinuity results to changing the weighting scheme, removing outliers and adding controls that should be continuous across borders. The first row reproduces the baseline. Row 2 runs the regressions (4) and (5) weighting each border piece equally, rather than weighting them by population. The estimates shrink relative to the baseline in magnitude, but remain statistically significant at least at 10 %. Hence, while population weighting is important to estimating the magnitude of growth discontinuities, it is not crucial for finding that statistically significant discontinuities exist. The subsequent row reports estimates in which the original population weights are truncated at their median value. The next row reproduces the baseline excluding the borders for which either the difference in growth rates at the border or the difference in growth rates between the two bordering countries exceeds in absolute value the 99th percentile of the distribution of the absolute values of these differences in the sample. The resulting numbers of observations and borders are reported below: we see that no borders (but over 200 observations) are excluded for the 5-year growth rates, while approximately seven borders are excluded for the 18-year growth rates. Removal of outliers does not substantially change the magnitudes of any of the coefficients (if anything, some of them increase slightly).

I now address the hypothesis that different data collection procedures on different sides of borders, or omitted spatially varying variables on different sides of borders, may be driving the growth discontinuities at borders. First, whenever growth in lights per capita is the dependent variable, the estimates depend on population data from the Gridded Population of the World (GPW) dataset, which reports population estimates for small subnational units that vary in size from country to country. In some countries, such units are larger than others, leading to coarser spatial population estimates, and thus, potentially, spurious discontinuities in growth rates at national borders. This concern is partially addressed by presenting estimates with growth in light density as a dependent variable; however, as a further check in Row 5 of Table 3 I include the log average number of people per subnational unit per country and the log spatial resolution of the subnational unit grid of each country (both from GPW) as right-hand-side variables. The estimates appear to be largely unaffected, with some standard errors increasing. Another concern may be that different border pieces have different areas, or that growth in nighttime lights mismeasures growth in output in a way that is correlated with population (for example, if shared public goods contribute to nighttime lights, then growth in lights per capita understates actual growth in output per capita). To address these concerns, in Row 5 I reproduce the baseline specification augmented by separate quartic polynomials in log total population on a given side of a border piece in a given year (which is a 70-km wide neighborhood of the border) and in the log area of each such side. Adding these controls does not substantially affect my estimates. Rows 6 and 7 add climatic variables (described in Sect. 4) and local public goods variables (described in Sect. 6.2) as controls, with only minor changes to the estimated discontinuities. Finally, Row 8 includes all controls from rows 5-7, as well as the fraction of the area of each border piece covered by oil and gas flares

Dep. Var.	(1) 5-Year growth	(2) 5-Year growth	(3) 18-Year growth lights p/c	(4) 18-Year growth lights p/c
Ind. Var.	Indicator	Elasticity	Indicator	Elasticity
Baseline estimates	1.48***	.38***	1.28**	.45**
	(.51)	(.11)	(.57)	(.20)
No weights	.73**	.22***	.56*	.23**
	(.30)	(.07)	(.32)	(.10)
Truncation of weights	.87**	.24***	.48	.24**
	(.34)	(.08)	(.32)	(.10)
No outliers	1.56***	.43***	1.32**	.51**
	(.49)	(.13)	(.55)	(.22)
GPW resolution Ctrls	1.58**	.40***	1.42**	.49*
	(.61)	(.14)	(.65)	(.26)
Population and area	1.33***	.35***	1.13**	.39**
	(.42)	(.11)	(.44)	(.16)
Climate controls	1.34***	.36***	1.25***	.47***
	(.42)	(.10)	(.42)	(.15)
Local public goods Ctrls.	1.51***	.37***	1.33***	.45***
	(.42)	(.10)	(.50)	(.17)
All controls	1.32***	.35***	1.34***	.46***
	(.38)	(.10)	(.39)	(.13)
Border-year fixed effects	Yes	Yes	Yes	Yes
No. observations	5178	5178	1290	1290
No. borders	270	270	264	264
R^2 of first row	.02	.02	.07	.06

 Table 3 Growth border discontinuities with controls

Data on lights and population available from the NOAA and CIESIN, respectively, for 1990 (1992 for lights), 1995, 2000, 2005 and 2010. Each column presents estimates of Eq. (4) or (5). Observation unit is a countryborder piece-year. Robust standard errors clustered on border and taking into account two-step estimation and infill asymptotics in parentheses. Each observation weighted by total population in the respective buffer piece, except in the no-weight specification. The dependent variable in each specification is the Eq. (3) local linear estimate of the indicated concept of growth of lights at the border corresponding to the given country-border piece year. The independent variable, whose coefficient is given in each cell, is the indicator that a given side of the border corresponds to the country with the higher corresponding national growth rate (5-year or 18-year) of GDP per capita from WDI, or the corresponding national growth rate itself. For the no-outlier specification, border piece-years with differences in growth of lights at border or differences in nationwide growth of GDP per capita greater than the 99th percentile of the distributions of these variables in absolute value are removed. For the truncated weights specification, weights are truncated at the median of their distribution. The GPW controls are log average resolution of the subnational unit net for the country in question and log number of people per subnational unit. The population and area specification includes separate quartic polynomials in population and area of the 70-km neighborhood around each border. The climate and local public goods controls are described in Sect. 4

(DMSP-OLS). The discontinuity estimates are slightly lower than in the baseline, but their magnitudes and significance levels remain very comparable despite the addition of such an extensive battery of control variables.

4 Heterogeneity of growth discontinuities at the border

As mentioned in the Introduction, one immediate explanation for discontinuities in the growth rate of economic activity may be that national borders tend to be drawn at discontinuous changes in geographic variables, such as altitude, slope of the terrain (a related measure, ruggedness, was investigated in Nunn and Puga [2012]), and climatic zones. The historical process through which the border was set may also shape patterns of economic growth on either side. In this section, I investigate whether the physical and historical characteristics of borders may explain border discontinuities in economic growth. Overall, I find only weak evidence that preexisting geographical discontinuities or border history play an important role in explaining the large differences in growth rates found on average at national borders.

4.1 Geographical differences across borders

I calculate average values of altitude, slope, and statistics of temperature and precipitation (means, maximums, minimums, variability measures) on either side of every border, obtained from WorldClimate at 30 arcsecond resolution. I also calculate, on either side of every border, the fraction of land belonging to each of the 14 climatic zones defined by the International Geosphere Biosphere Programme, by using the Global Land Cover Characteristics Data available from USGS, also at 30 arcsecond resolution.¹⁴ First, in Tables 4 and 5, I investigate whether discontinuities in economic growth rates tend to be located on borders with large differences in some of these variables across the border, or whether they are also present at borders on which both sides are relatively similar in terms of these geographic variables. Specifically, I classify border pieces as having a difference in any particular geographical variable across the two sides of the border as above the median or below the median of such differences. I find that discontinuities in 5-year economic growth rates are present on border pieces with below-median border differentials in altitude, slope, temperature and its standard deviation, precipitation, fraction of land used as cropland, and fraction of land endowed with nutrients. Discontinuities in 18-year economic growth rates tend to be statistically insignificant on average on borders with below-median differentials in slope, in temperature and its standard deviation as well as in the fraction of land used as cropland. However, the point estimates of the 18-year economic growth rate discontinuities tend to be of the same order of magnitude on borders with below- as well as with above-median differentials in geographic covariates. Moreover, it is reassuring to see that borders with below-median differences in the fraction of land endowed with nutrients, a more exogenous measure of land suitability for agriculture than the fraction of land that is used as cropland, have statistically significant discontinuities in 18-year economic growth rates.

¹⁴ The climatic variables are: (1) Annual Mean Temperature, (2) Mean Diurnal Range, (3) Isothermality, (4) Temperature Seasonality, (5) Max Temperature of Warmest Month, (6) Min Temperature of Coldest Month, (7) Temperature Annual Range, (8) Mean Temperature of Wettest Quarter, (9) Mean Temperature of Driest Quarter, (10) Mean Temperature of Warmest Quarter, (11) Mean Temperature of Coldest Quarter, (12) Annual Precipitation, (13) Precipitation of Wettest Month, (14) Precipitation of Driest Month, (15) Precipitation Seasonality, (16) Precipitation of Wettest Quarter, (17) Precipitation of Driest Quarter, (18) Precipitation of Warmest Quarter, (19) Precipitation of Coldest Quarter. The land cover categories are: (0) Body of Water, (1) Evergreen Needleleaf Forest, (2) Evergreen Broadleaf Forest, (3) Deciduous Needleleaf Forest, (4) Deciduous Broadleaf Forest, (5) Mixed Forest, (6) Closed Shrublands, (7) Open Shrublands, (8) Woody Savannas, (10) Grasslands, (11) Permanent Wetlands, (12) Croplands, (13) Cropland / Natural Vegetation Mosaic.

	•								
Border Type	(I) All	(2) Alt jump	(3) No Alt jump	(4) Slope jump	(5) No Slope jump	(6) Temp jump	(7) No Temp jump	(8) TempSD jump	(9) No TempSD jump
Dummy, 5-Year growth	1.48^{***}	1.27*	1.70^{***}	1.87^{***}	1.06^{**}	1.84^{***}	1.11*	1.91^{**}	1.01*
Lights per capita	(.51)	(.74)	(.50)	(.72)	(.48)	(89)	(.63)	(.75)	(09)
5-Year growth	.38***	.38**	.38***	.52**	.28**	.54***	.22	.40**	.36***
Lights per capita	(.11)	(.18)	(.13)	(.20)	(.11)	(.17)	(.14)	(.19)	(.13)
Dummy, 18-Year growth	1.28^{**}	67.	1.77^{***}	1.92^{**}	.57	1.68^{**}	.86	1.55*	.97
Lights per capita	(.57)	(.79)	(.58)	(.85)	(.42)	(99)	(.78)	(.86)	(.64)
18-Year growth	.45**	.14	.73***	.70**	.22	.64***	.25	.54	.37
Lights per capita	(.20)	(.31)	(.21)	(.35)	(.17)	(.23)	(.29)	(.36)	(.23)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	622	730	562	190	666	686	586	766
No. borders	270	190	179	183	210	194	175	180	188
R^2 of first row	.04	00 [.]	.16	.08	.01	.08	.01	.06	.02
Data on lights and popula from SRTM. Data on clim a border piece exceeds th Robust standard errors clu respective buffer piece. Th he given country-border r with the higher correspore	tion available atic variables population-v stered on boru e dependent biece year. Th	from the NC available fro weighted mec der and taking variable in ea te independer growth rate (AA and CIESIN m WorldClimate lian of these diffe into account two ch specification i t variable, whosa 5-vear or 18-veas	I, respectively, (Hijmans et al. prences. Each c o-step estimatic s the Eq. (3) loc e coefficient is of GDP per c	for 1990 (1992 for 2005). A "jump" is column presents est an and infill asympt cal linear estimate given in each cell, a apita from WDL of	lights), 1995, 2 i defined to take imates of Eq. (6 otics in parenth of the indicated is the indicator the correspond	000, 2005 and 201 place whenever th th or (5). Observati esss. Each observati concept of growth that a given side of thing national growth	0. Data on altitude e difference in the. on unit is a countr ion weighted by to of lights at the borrde the border correst h rate itself	and slope obtained climatic variable for y-border piece-year. tal population in the ler corresponding to onds to the country
	•	,		1			,		

 Table 4
 Growth discontinuities by border type

Dowlose Three	(I)	(2) Decoira inmo	(3) No Decin inner	(4) E	(5) No Erro Actimus	(6) Nutrii intriv	(7) No Mittei inmo
poinci type	III	ricup. Jump	TNO LICOP. Jump	TIACAS Jump	duing Brant uni	diring tinuri	
Dummy, 5-Year growth	1.48^{***}	1.38*	1.58***	1.85^{***}	1.11^{*}	1.86^{***}	1.09*
Lights per capita	(.51)	(.74)	(.58)	(09.)	(09.)	(.67)	(.62)
5-Year growth	.38***	.30**	.48***	.70***	.17	.42**	.35**
Lights per capita	(.11)	(.15)	(.15)	(.18)	(.11)	(.18)	(.14)
Dummy, 18-Year growth	1.28^{**}	.95	1.62^{**}	1.94^{***}	.63	1.65*	.89
Lights per capita	(.57)	(.67)	(.63)	(.68)	(.63)	(.85)	(.63)
18-Year Growth	.45**	.27	.61***	.72**	.23	.55	.37*
Lights per capita	(.20)	(.28)	(.22)	(.32)	(.21)	(.36)	(.22)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	782	570	430	922	496	856
No. borders	270	213	165	127	203	144	217
R^2 of first row	.04	.01	.10	.11	.01	60.	.02
Data on lights and populatio from SRTM. Data on climati a border piece exceeds the p Robust standard errors cluste respective buffer piece. The the given country-border piec	n available from c variables avails opulation-weigh red on border an lependent variab ce vear. The inde	the NOAA and CIE: able from WorldClim. ted median of these c d taking into account de in each specificatic ependent variable. wh	SIN, respectively, for 19 state (Hijmans et al. 2005) lifferences. Each column two-step estimation and on is the Eq. (3) local line lose coefficient is given 1	90 (1992 for lights), 1 A "jump" is defined t presents estimates of infill asymptotics in p in actimate of the indi in each cell, is the indi in each cell, is the indi	995, 2000, 2005 and 201 to take place whenever th Eq. (4) or (5). Observati, arentheses. Each observati cicated concept of growth of icator that a given side of	 Data on altitude: a difference in the cl or unit is a country. ion weighted by tota the border correspondent 	and slope obtained imatic variable for border piece-year. Il population in the ar corresponding to onds to the country
with the higher correspondin	g national growt	th rate (5-year or 18-y	/ear) of GDP per capita f	from WDI, or the corr	esponding national growt	h rate itself	,

Even though there is some evidence that 18-year economic growth rates may be discontinuous mostly at borders with geographic differentials, it does not appear that these geographical differentials are driving the finding of border discontinuities in economic growth. First, I include all the geographic and climatic variables as covariates on the right-hand side in the regressions (4) and (5). In Row 5 of Table 3, I present the resulting discontinuity estimates. Including these controls weakens most of the estimated coefficients only slightly and increases a few. Second, I rerun the specifications given by Eq. (4) replacing the dependent variable by each of the 35 geographic variables used. Out of 70 possible tests, 2 reject at 5 %, which is below the number that one would expect if none of the geographic variables were discontinuous (which is about 4). I present the local linear regressions for the geographic variables thought to be potentially important for economic growth (and for which the type of border breakdowns were computed in Tables 4 and 5) in Table 6: altitude, slope, mean temperature, temperature standard deviation, mean precipitation, fraction of land that is cropland and fraction of land with good nutrient availability. All the variables are scaled to have mean zero and standard deviation equal to unity, and for all variables presented, the discontinuity estimate is not significant. It should be noted that the climatic (though not the altitude and slope) variables are interpolated from weather stations, and the land cover variables are categorical, making it less surprising that we do not find discontinuities in them. Both these checks imply that there is no systematic tendency for sides of borders corresponding to faster-growing countries to have higher or more volatile temperatures, or to be more rugged than the sides of borders corresponding to slower-growing countries.

4.2 Discontinuities by continent

It is important to understand which borders contribute most to my baseline finding, and whether particular clusters of borders drive my results. Table 7 begins by presenting estimates of border discontinuities for the world excluding particular regions to ensure that no region is driving my main result. It is apparent that the estimates are similar across regions. Subsequently, Table 7 presents estimates of discontinuities in the growth rates of lights per capita at borders between OECD countries, between post-Communist countries and other (potentially not post-Communist) countries, in Asia, Africa and in the Americas.¹⁵ The borders with the strongest discontinuity estimates are the borders of post-Communist countries. Growth discontinuities at African borders, though typically not statistically significant have a similar magnitude as the baseline growth discontinuity estimates.

4.3 Discontinuities by type of border

It is instructive to compare discontinuities in economic growth across partitions of the set of borders that are other than continents. Table 8 presents estimates of growth discontinuities for borders that differ in age, artificiality and remoteness from the bordering countries' heartlands. Column 1 reproduces the baseline estimates. Columns 2 and 3 contrast growth discontinuities across recently established borders—those that have become national borders since the collapse of the Soviet Union, Czechoslovakia and Yugoslavia—with growth discon-

¹⁵ The Americas are treated as a single continent because they are contiguous, and the number of borders in the Americas is comparable to that in Europe, Asia or Africa. Since the latter three continents form a continuous landmass, I define Russia, Turkey and Egypt to be part of two continents at once. I count the borders of these countries to belong to the continent of the bordering country: e.g. Russia's border with Estonia is a European border, but Russia's border with China is an Asian border.

Dep. Var.	(1) Log population	(2) Log altitude	(3) Log slope	(4) Mean temperature	(5) Log SD Temp.	(6) Log precipitation	(7) Fraction cropland	(8) Fraction nutrients
Dummy, 5-Year GDP growth nationwide	05	.01	.08	01	00	00	.05	.08
	(.03)	(.02)	(90.)	(.01)	(00)	(.00)	(90)	(80)
Dummy, 18-Year GDP growth Nationwide	06	.02	60.	01	00	00.	.11	.11
	(.05)	(.03)	(90.)	(10.)	(00)	(.01)	(.10)	(.14)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	1352	1352	1352	1352	1352	1352	1352
No. borders	270	270	270	270	270	270	270	270
R^2 of last row	.01	.01	.07	.02	.01	2.20	00.	.00
Data on lights and population available from (Hijmans et al. 2005). All climate variables n growth. Observation unit is a country-border weighted by total population in the 70 -km ne geographical control variable at the border co that a over side of the border correstonds to	n NOAA and CIH escaled to have n piece-year. Robu ighborhood arouu rresponding to th	SSIN. Data or nean 0 and van 1st standard er nd the buffer F e given counti	1 altitude an iance 1. Eac rors cluster viece. The de y-border pie	d slope obtained fi chorn presents ed on border and te pendent variable il ce year. The indep	rom SRTM. Dat estimates of Eq. king into accour n each specificat endent variable,	a on climatic variab (4) or (5) with clim at infill asymptotics ion is the Eq. (3) loc.	les available from aate variables repla in parentheses. Ea al linear estimate of given in each cell,	WorldClimate cing economic ch observation of the indicated is the indicator

J Econ Growth (2017) 22:145-192

Table 7 Growth discontin	nuities by c	ontinent									
Continent	(1) All	(2) Not OECD	(3) Not post- communist	(4) Not Asia	(5) Not Africa	(6) Not America	(7) Only OECD	(8) Only Post Communist	(9) Only Asia	(10) Only Africa	(11) Only America
Dummy, 5-Year growth	1.48^{***}	1.60^{***}	1.35^{**}	1.43^{**}	1.42^{**}	1.56***	03	2.54***	1.53*	1.66*	.58
Lights per capita	(.51)	(.53)	(.58)	(.57)	(09)	(.53)	(.45)	(.85)	(.84)	(1.00)	(1.20)
5-Year growth	.38***	.38***	.34**	.35**	.46***	.38***	11	.60***	.42**	.25	.54
Lights per capita	(.11)	(.12)	(.13)	(.14)	(.14)	(.12)	(.63)	(.19)	(.19)	(.19)	(.61)
Dummy, 18-Year growth	1.28^{**}	1.41^{**}	1.28^{**}	.98*	1.22*	1.35^{**}	57**	1.21*	1.58*	1.43	.48
Lights per capita	(.57)	(.59)	(.62)	(.57)	(69)	(.61)	(.24)	(.64)	(.92)	(1.00)	(.91)
18-Year growth	.45**	.45**	.43*	.48**	.47	.43**	64	.62**	.41	.41	1.12
Lights per capita	(.20)	(.20)	(.22)	(.21)	(.28)	(.21)	(.55)	(.26)	(.34)	(.27)	(69)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	1230	1092	1018	914	1154	122	260	334	438	198
No. borders	270	247	214	206	178	235	23	56	64	92	35
R^2 of first row	.04	.04	.03	.04	.05	.04	.03	.11	.06	.03	.07
Data on lights and popula Eq. (4) or (5). Observation parentheses. Each observa indicated concept of grown indicator that a given side corresponding national gro	tion availat n unit is a c tion weight th of lights of the bord owth rate its	le from the NC ountry-border J ed by total pop at the border co ler corresponds self. The defini	DAA and CIE piece-year. Rc ulation in the presponding t to the countr tions of the re	SIN, respectiv obust standard respective but to the given cc y with the hig gions are expl	rely, for 1990 errors clustere ffer piece. The ountry-border J her correspond ained in Sect.	(1992 for lighted on border of dependent variation of the dependent variation of the dependent variation of the dependent of	tts), 1995, 20 and taking ini uriable in eacl in independer growth rate (00, 2005 and 20 to account two-s a specification is the variable, whose 5-year or 18-yea	10. Each cc tep estimati the Eq. (3) e coefficien r) of GDP _I	olumn present on and infill <i>i</i> or local linear e t is given in e: oer capita fror	s estimates of symptotics in stimate of the tch cell, is the n WDI, or the

Border type	(1) All	(2) Old (Pre-1992)	(3) New (Post-1992)	(4) Squiggly	(5) Straight	(6) Rugged	(7) Flat	(8) Near	(9) Far	(10) Has Flares	(11) No Flares
Dummy, 5-Year growth	1.48^{***}	2.45**	1.42***	1.25	1.72^{***}	96.	1.99^{***}	1.52^{**}	1.44	52	1.72^{***}
Lights per capita	(.51)	(1.15)	(.54)	(.88)	(.61)	(.65)	(69.)	(.62)	(.88)	(.95)	(.52)
5-Year growth	.38***	.35*	.38***	.29	.43***	.39**	.37**	.31**	.49**	12	.44**
Lights per capita	(.11)	(.18)	(.13)	(.24)	(.13)	(.16)	(.18)	(.14)	(.20)	(.29)	(.12)
Dummy, 18-Year growth	1.28^{**}	.81	1.30^{**}	1.47	1.08*	1.26^{**}	1.29	66.	1.57	50	1.49^{**}
Lights per capita	(.57)	(.86)	(.59)	(.92)	(.59)	(.61)	(1.00)	(.62)	(96.)	(.73)	(09)
18-Year growth	.45**	.38	.45**	.64	.34*	.43**	.46	.36	.53	49	.55**
Lights per capita	(.20)	(.30)	(.22)	(.42)	(.18)	(.21)	(.43)	(.25)	(.35)	(.32)	(.22)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	152	1200	450	902	580	772	512	840	236	1116
No. borders	270	27	243	110	160	151	176	143	163	75	220
R^2 of first row	.04	.06	.04	60.	.02	.04	.03	.02	.05	.04	.06
Data on lights and popula Eq. (4) or (5). Observation parentheses. Each observa indicated concept of growt	tion availabl 1 unit is a co- tion weighte th of lights al	e from the NOAA a untry-border piece-y d by total populatio t the border correspo	und CIESIN, respectiv year. Robust standard n in the respective but anding to the given co	ely, for 1990 errors cluste ffer piece. Th untry-border) (1992 for 1 red on bords the dependent	ights), 1995 er and taking t variable in The indeper	, 2000, 2005 g into accour each specifi ident variabl	i and 2010. It two-step cation is the e, whose cc	Each col estimatio e Eq. (3)]	lumn presents on and infill asy local linear est is given in eac	estimates of mptotics in imate of the h cell, is the

 Table 8
 Growth discontinuities by border type

indicator that a given side of the border corresponds to the country with the higher corresponding national growth rate (5-year or 18-year) of GDP per capita from WDI, or the

corresponding national growth rate itself. The definitions of the regions are explained in Sect. 4

tinuities across all other borders, which typically date at least to the decolonization period of the 1960s. The sample of newly established borders is very small, but because the ex-republics of the former Soviet Union had very different growth experiences after independence, there is good reason to believe that it may be illustrative. In fact, we do observe discontinuities in growth rates at the border for both the old and the new borders, with the magnitudes of the discontinuities roughly similar (although the discontinuities across the newly established borders are not always precisely estimated). Finding similar growth discontinuities across new borders could be consistent with growth discontinuities arising from long-term processes that may also be endogenous to border formation.

It is also interesting to ask whether growth discontinuities are similar across borders that have been drawn artificially and borders that appear to follow some natural geographic formation. Finding no discontinuities across artificially drawn borders may be concerning for the same reason as it would be to find no growth discontinuities across recently formed borders: it would suggest that the endogeneity of border creation might play a part in whether there is a growth discontinuity across the border or not. To measure border artificiality, I use an index of fractal dimension that is similar to the measure of Alesina et al. (2011), and compute separate fractal indices for all borders in my dataset.¹⁶ I break up my sample of borders into two at the border population-weighted median of the fractal index; the borders with higher fractal index I refer to as "squiggly" and the borders with lower fractal index I refer to as "straight." Columns 4 and 5 show that, if anything, growth discontinuities are somewhat stronger at straight borders than at squiggly borders, though this varies by the dependent variable used to measure growth discontinuities at borders, and all of the growth discontinuity estimates are fairly close to their baseline values. Another way of partitioning natural and artificial borders is to consider whether the border is highly rugged,¹⁷ which would suggest that it is a natural border. Using the ruggedness measure based on WorldClimate data from Sect. 3, I partition border pieces into two groups:¹⁸ those with average ruggedness in a 70-km strip around the border higher than the border population-weighted median of this measure, and those with lower average ruggedness. Columns 6 and 7 show that the border discontinuity estimates are similar to each other and to the baseline estimates in magnitude for rugged as well as for flat borders (this similarity is particularly evident for regressions on differences in national growth rates rather than for regressions on indicators).

Michalopoulos and Papaioannou (2014) find that discontinuities in economic activity at a border that bisects an ethnic group are low for African borders far away from the bordering countries' capitals and high for African borders close to the bordering countries' capitals,

¹⁶ The fractal dimension of a curve is computed by superimposing square grids of varying grid sizes over the curve, counting up the number of grid cells intersected by the curve for each size, and running a regression of the log number of intersections on log grid size. The negative of the slope coefficient (minus one) is the fractal index of the border. See Alesina et al. (2011) (AEM) for greater details.

I do not succeed in replicating the results of AEM, most likely because of the worse resolution of the border map that I have available to use. The fractal dimension indices for countries that I obtain have a statistically significant correlation with AEM's of 0.55. In results not reported, I use an alternative method of approximating the fractal dimension of borders by simply taking the average of the fractal dimensions of the bordering countries. I obtain the same results when I use this second method.

¹⁷ In this paper, I use the term "rugged" to mean that terrain has high average slope. I note that this is somewhat different from the concept of ruggedness in Nunn and Puga (2012), but is close in spirit.

¹⁸ Since I have ruggedness for every border piece, I can allow a border to have both rugged and flat border pieces. Hence, the numbers of borders having rugged and flat border pieces do not need to add up to the total number of borders (though they come fairly close, suggesting that most borders are either completely rugged or completely flat).

and explain this finding by noting that distance to the capital is strongly associated with the extent of the reach of the state (in Africa). It therefore is interesting to ask whether border discontinuities around the world may be stronger for borders near the bordering countries' capitals. Hence, I divide the border pieces sample into border pieces near to the bordering countries' capitals and border pieces far from the bordering countries' capitals. (It makes sense to do this at the border piece level because, for example, the Chile-Argentina border is closer to the two countries' capitals in the north than it is in the south, so it is probably incorrect to treat all of it as being at the same distance to their capital as it is in the north). Each border piece whose maximum distance to a bordering countries' capitals, and the rest are classified as far from the bordering countries' capitals. It once again does not appear that there is a noticeable difference in the strength of growth discontinuities across these groups of border pieces; the magnitudes of the discontinuity estimates being similar for border pieces near their capitals and border pieces far from capitals.

Cust and Harding (2014) document that when oil fields straddle borders, investors choose to drill for oil on the better institutionalized side. It is interesting to examine the extent to which differential fossil fuel extraction explains growth discontinuities at borders more generally. Using the dataset of gas flares provided by DMSP-OLS,¹⁹ I partition border pieces according to whether they contain gas flares, which are good indicators of fossil fuel extraction activity going on at the border.²⁰ We observe that restricting the sample only to border pieces without flares yields results very similar to the baseline, while on the sample of border pieces with flares, the discontinuities are typically statistically insignificant. This analysis suggests that gas flares are not responsible for growth discontinuities at borders. The fact that there are no growth discontinuities at borders with gas flares is intuitive, because flares propagate continuously across space and blur any discontinuity that may otherwise be observed. There are also relatively few borders with gas flares: only 236 of the 1352 border pieces in our analysis have gas flares.

5 Barriers to economic activity across borders

5.1 Growth discontinuities and economic barriers

As I have discussed in the introduction, if border location does not appear to explain discontinuities in economic growth at borders, then the existence of growth discontinuities at borders is consistent with certain variables that are determinants of growth (such as institutions or culture) being different on different sides of borders and the border acting as a barrier to their diffusion. Growth discontinuities are also consistent with the determinants of growth on both sides of the border being the same, but the border acting as a barrier to the spread of economic shocks. In this section, I will explore the second hypothesis. I will find some evidence that measures of the extent to which borders act as barriers to trade and economic

¹⁹ The gas flares data are available country by country from the DMSP-OLS website. I am very grateful to Matthew Lowe (http://economics.mit.edu/grad/mlowe) for sharing his version of these datasets merged together.

²⁰ Clearly, the gas producing the flares is wasted rather than captured, and the extent to which flares are produced depends on how lax environmental regulations are, so the gas flares do not measure the quantity of fossil fuel extraction. However, it is reasonable to believe that the presence of flares should indicate that at least some fossil fuel extraction activity is taking place.

exchange, and therefore to economic shocks, do not affect the size of growth discontinuities at these borders.

It is well-known that national borders act as discontinuous increases in transaction costs for trade (McCallum 1995). Even in a world with countries having identical institutions, public goods provision, culture and other determinants of economic growth, we could then still see discontinuities in economic activity at borders because local economic shocks would not transmit to neighboring countries but would remain contained in the country of origin. If borders were completely impermeable to economic exchange such as trade, investment and migration, different market equilibria would prevail in each country, and shocks that affect one country's equilibrium would have no effect on its neighboring country, thus creating a discontinuity in prices, wages, quantities, and most likely economic growth across the countries' common border.

I test whether this transaction costs channel is important by looking at whether the presence of economic exchange—trade, migration or capital—mitigates discontinuities in economic activity at borders. Since the transaction costs channel should manifest itself through differences in the cross-border flows mentioned above, if the extent of cross-border flows across the border (suitably normalized to account for some of their obvious determinants) does not affect whether or not growth discontinuities are present at borders, then it is unlikely that borders pose sufficient barriers to economic exchange to explain the existence of growth discontinuities. Instead, different growth determinants, such as different institutions or culture, must prevail on different sides of borders, with the border preventing them from directly influencing growth on the other side.

I use four proxies of cross-border flows: (1) total trade in goods and services in 2000, normalized by the product of the trading countries' GDPs, (2) total migration flows in 2011, normalized by the product of the countries' populations, (3) average tariff levels across each border in 2000, and (4) total remittances between the bordering countries in 2011, normalized by the product of their GDPs. The trade variable is from IMF's Direction of Trade Statistics, while the other three variables are from the World Bank.

A direct way to look at the association between border pemeability and the magnitudes of growth discontinuities at borders is simply to look at whether borders with high crossborder flows have smaller differences in growth rates at the border. To do so, I run the OLS regression:

$$\left|\Delta \tilde{g}_{i,b,t}\right| = \alpha + \delta T_b + \varepsilon_{i,b,t} \tag{8}$$

where $|\Delta \tilde{g}_{i,b,t}|$ are absolute values of growth differentials, and α and δ are coefficients to be estimated.²¹ The variable T_b is one of the measures of cross-border flows described above. It is scaled to have mean zero, unit variance, and to have larger values of T_b imply stronger cross-border flows (hence, the trade barriers variable would be multiplied by -1). If growth discontinuities exist at borders because the borders are barriers to the propagation of economic shocks, then the coefficient δ should be negative and statistically significant.

Additionally, we would expect borders with stronger cross-border flows to have a weaker relationship between growth at the border and growth nationwide. To measure this, I augment the regressions in Sect. 3 by an interaction of a cross-border flow measure with the fastergrowing country indicator or the gap in the growth rates of the bordering countries. The regressions from Sect. 3 then become

²¹ It does not make sense to regress growth differentials directly on the border permeability measures because there is nothing in the regression that would explain the sign of the growth differential: the constant and border permeability measures can only explain the magnitude of the differential.

$$\tilde{g}_{i,b,t} = \alpha_{b,t} + \beta u_{i,b,t} + \beta u_{i,b,t} \times T_b + \varepsilon_{i,b,t}$$
(9)

$$\tilde{g}_{i,b,t} = \chi_{b,t} + \gamma g_{i,t} + \tilde{\gamma} g_{i,t} \times T_b + \upsilon_{i,b,t}$$
(10)

where $\tilde{\beta}$ and $\tilde{\gamma}$ are coefficients on the interaction term of one of the cross-border measures with either the faster-growing country indicator or with the national growth rate of a country. All other variables are as defined in Sect. 3.²² The idea behind these regressions is that if the absence of cross-border flows explains growth discontinuities, then borders with more cross-border exchange should exhibit a weaker relationship between differences in growth rates at the border and differences in growth rates between bordering countries. Therefore, $\tilde{\beta}$ and $\tilde{\gamma}$ should be negative and large, so that for borders with larger values of T_b , growth discontinuities are smaller.²³ All the border permeability variables are obviously at the border level, and hence are captured by border fixed effects, so I can use them only to create interaction terms in regressions.²⁴

I present results for the specifications (8), (9) and (10) in Table 9. For each cross-border flow measure, Table 9 presents first the results from the univariate regression (8), and then from the interaction specifications (9) or (10). The univariate regression results suggest that trade barriers are low (given the sign convention) on borders with high discontinuities, and that migration is high across borders with large discontinuities (which is not surprising, since people might be willing to migrate only if their income at the destination is likely to grow

$$\left|\Delta \tilde{g}_{i,b,t}\right| = \alpha + \sum_{j=1}^{4} d_j T_b^j + \varepsilon_{i,b,t}$$
(11)

and

$$\tilde{g}_{i,b,t} = \alpha_{b,t} + \beta u_{i,b,t} + \sum_{j=1}^{4} \tilde{\beta}_j u_{i,b,t} \times T_b^j + \varepsilon_{i,b,t}$$

$$\tilde{g}_{i,b,t} = \chi_{b,t} + \gamma g_{i,t} + \sum_{j=1}^{J} \tilde{\gamma}_j g_{i,t} \times T_b^j + \upsilon_{i,b,t}$$
(12)

Estimates from these equations are presented in Appendix Table A1.

Additionally, I perform sample splits of the data along borders with high and low cross-border flows according to the four measures I consider. Appendix Table A3 presents estimates from this approaches. The results are qualitatively similar, although it appears that growth discontinuities may be weaker at borders with above-median trade flows for some measures of growth. The interaction-based specification is preferable to the sample split because it allows the extent of cross-border flows to have a more continuous effect on the size of the border discontinuity.

²³ Following Sect. 3, we can also rewrite these equations as

$$\Delta \tilde{g}_{i,b,t} = \beta \Delta u_{i,b,t} + \beta \Delta u_{i,b,t} * T_b + \Delta \varepsilon_{i,b,t}$$

and

$$\Delta \tilde{g}_{i,b,t} = \gamma \Delta g_{i,t} + \tilde{\gamma} \Delta g_{i,t} * T_b + \Delta \upsilon_{i,b,t}$$

to highlight that the border permeability terms should attenuate the effect of the national growth differences upon the growth differences at the border.

²⁴ This approach is the same as performing a sample split of the data along borders with high and low cross-border flows, with the exception that interactions allow the "split" variable to be continuous.

²² I can also compute versions of these regressions in which all the cross-border flow measures are included simultaneously. This approach has the advantage of capturing the effect of controlling for a broader measure of cross-border flows on growth discontinuities at borders, although the coefficients on the interaction effects become more difficult to interpret. The regressions then become

	(1)	(2)	(3)	(4)
Dep. Var.	5-Year growth p/c	5-Year growth p/c	18-Year growth p/c	18-Year growth p/c
Ind. Var.	Dummy	Elast.	Dummy	Elast.
Barrier Measure: Trade Volume	.52 (.39)	.52 (.39)	27 (.20)	27 (.20)
R^2	.04	.04	.00	.00
Nat. Output Measure	1.48***	.37***	1.27**	.45**
*	(.49)	(.11)	(.54)	(.20)
Output Measure X Barrier Measure	31	08	47	.00
	(.43)	(.07)	(.48)	(.12)
R^2	.04	.04	.00	.00
Barrier Measure: Trade Barriers	45	45	55***	55***
	(.45)	(.45)	(.18)	(.18)
R^2	.04	.04	.02	.02
Nat. Output Measure	1.48***	.47***	1.28**	.60**
	(.51)	(.16)	(.56)	(.24)
Output Measure × Barrier Measure	07	.14	.13	.29*
	(.32)	(.11)	(.31)	(.17)
R^2	.04	.04	.02	.02
Barrier Measure: Remittances	21	21	05	05
	(.43)	(.43)	(.27)	(.27)
R^2	.04	.04	.00	.00
Nat. Output Measure	1.46***	.45***	1.26***	.62**
	(.41)	(.16)	(.47)	(.26)
Output Measure X Barrier Measure	.91**	.12	.89*	.29
	(.44)	(.10)	(.48)	(.18)
R^2	.04	.04	.00	.00
Barrier Measure: Migration	.22	.22	.23	.23
	(.35)	(.35)	(.24)	(.24)
R^2	.04	.04	.00	.00
Nat. Output Measure	1.48***	.37***	1.28**	.44*
	(.51)	(.12)	(.57)	(.22)
Output Measure X Barrier Measure	05	05	08	02
	(.29)	(.06)	(.33)	(.11)

Table 9 Cross-border flows

Table 9 continued

	(1)	(2)	(3)	(4)
<i>R</i> ²	.04	.04	.00	.00
Border-Year Fixed Effects	Yes	Yes	Yes	Yes
No. Observations	5408	5408	1352	1352
No. Borders	270	270	270	270

Data on lights and population available from the NOAA and CIESIN, respectively, for 1990 (1992 for lights), 1995, 2000, 2005 and 2010. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account two-step estimation and infill asymptotics in parentheses. Each observation weighted by total population in the respective buffer piece. For each cross-border flow measure, the top line presents the results from regression (8), while the bottom two lines present results from regressions (9) or (10). Data on trade is from IMF's Direction of Trade Statistics for 2000, data on trade barriers for 2011 is from the World Bank. Trade volume and remittances are normalized by product of bordering country GDPs, while migration is normalized by the product of the bordering countries' populations. All border permeability variables standard errors clustered on border and taking into account infill asymptotics in parentheses

much faster than their current income), but the estimates for remittances and trade volume do not have a consistent sign. For the interaction specifications, the coefficients β and γ remain approximately the same size and significance as in Table 2, while the coefficients on the interaction terms are either positive (suggesting that cross-border flows take place across borders with large discontinuities) or are negative but statistically insignificant and with small absolute value (so that a value that is 2 standard deviations above the mean for any of the cross-border flow measures would produce an effect double the size of the coefficient and would not nullify the discontinuity associated with the difference in national growth rates between the bordering countries). The signs of the interaction coefficients in the interaction specifications also do not match the signs of the coefficients in the univariate regression, which suggest that remittances tend to flow more across borders with smaller discontinuities, while trade and migration tends to flow more across borders with larger discontinuities. Since the extent of the specific cross-border interactions studied here does not predict heterogeneity in border discontinuities, I conclude that border discontinuities are unlikely to arise from the imperfect propagation of stochastic shocks to growth from one side of the border to the other. However, this conclusion depends on the mechanisms discussed in this section (mostly relating to trade and migration) being the relevant measures of border permeability, and it could be that other flows (such as of ideas and technologies, which may be important, though difficult to measure) may be true measures of barriers to the propagation of economic shocks at borders.

5.2 Growth discontinuities and cultural barriers

Another explanation for growth discontinuities at borders could be cultural differences between countries. For example, Algan and Cahuc (2010) find strong effects of trust on economic well-being of ethnic groups in the United States and Gorodnichenko and Roland (2016) find that differences in individualism (instrumented by the extent to which the country's blood type profile differs from that of the US or the UK) helps explain cross-country income differences. Tabellini (2010) also finds that trust, instrumented by historical literacy

rates and historical political institutions, has a positive effect on growth. While a national culture may arise through government policy (e.g. through inculcating certain values in national curricula or state-owned mass media), the mere fact that there exist barriers to cross-border interactions suggests that people in two different countries will be in different social networks and therefore will tend to develop different cultures. If culture indeed has direct effects on economic growth, such convergence to different cultural equilibria on different sides of national borders could lead to discontinuities.²⁵

To measure cultural similarity across borders, I obtain data on the genetic distance between populations of different countries from Spolaore and Wacziarg (2009). I also obtain data on whether the bordering countries share a common official language, whether the two countries were ever under the same colonizer, and the average distance between two randomly chosen people in the two bordering countries, all from Mayer and Zignago (2011). Using these variables, I re-run Eqs. (9) and (10), with cultural similarity variables in place of the border permeability measures T_b . I normalize the cultural similarity variables in the same way as I did the border permeability measures in the previous section (except the indicator variables), so negative coefficients mean that countries with greater cultural similarities have smaller growth discontinuities at borders. Table 10 presents the results. The magnitudes of the baseline discontinuity estimates (the main effects β and γ) decline slightly. For genetic distance and for population-weighted distance, the interaction coefficients tend to be negative, and are statistically significant in several specifications, suggesting that borders with large cultural barriers have smaller discontinuities. However, their magnitudes are only large enough to overwhelm the main effect for borders with cultural similarities being two or more standard deviations above the mean, hence only for the upper tail of national borders. For the same official language and same colonizer variables, the interaction coefficients generally are statistically insignificant, smaller in magnitude relative to the main effect, or positive. Hence, cultural barriers may play a greater role in explaining border discontinuities than the economic variables of the previous subsection.²⁶

6 Determinants of growth and border discontinuities

6.1 Correlates of growth discontinuities

Numerous factors have been hypothesized to explain differences in cross-country growth rates. One important hypothesis that I will investigate is that growth discontinuities at borders are explained, at least in part, by institutional differences between bordering countries. A large literature (going back to Adam Smith's *Wealth of Nations*) has identified institutions such as the rule of law (or the security of property rights) as important for economic growth and many papers have used it to proxy for institutional development more generally: notably, Acemoglu et al. (2001, 2005, 2014); Rodrik et al. (2004), and Michalopoulos and Papaioannou (2014). Acemoglu et al. (2001, 2002a, 2003) argue that institutions of private property are more important than geography, culture (proxied by religion and colonizer iden-

²⁵ Another way in which culture may interact with borders is if an ethnic group with the same culture is split by a border and made part of two different countries. The estimation technique that I use should difference out this common cultural component on both sides of borders. This subject has been explored at greater length in Michalopoulos and Papaioannou (2014).

²⁶ Analogously to the cross-border flow analysis, I compute estimates for the all-at-once regression and for using sample split. They are presented in Appendix Tables A2 and A4. The results are generally consistent with a weaker role for cultural variables in explaining border discontinuities than is Table 10.

	(1)	(2)	(3)	(4)
Dep. Var.	5-Year growth p/c	5-Year growth p/c	18-Year growth p/c	18-Year growth p/c
Ind. Var.	Dummy	Elast.	Dummy	Elast.
Cultural barrier:	52	52	19	19
genetic distance	(.37)	(.37)	(.27)	(.27)
R^2	.04	.04	.00	.00
Nat. Output Measure	1.44***	.36***	1.24**	.41**
	(.51)	(.11)	(.57)	(.21)
Output measure × cultural barrier	78**	22***	57	25*
	(.36)	(.07)	(.35)	(.15)
R^2	.04	.04	.00	.00
Cultural barrier: same official language	42	42	.69	.69
	(1.09)	(1.09)	(.64)	(.64)
R^2	.04	.04	.00	.00
Nat. output measure	1.79***	.40***	1.65**	.60**
	(.60)	(.13)	(.69)	(.23)
Output measure × cultural barrier	83	05	98	45
	(1.01)	(.28)	(1.04)	(.40)
R^2	.04	.04	.00	.00
Cultural barrier: same colonizer	-1.41*	-1.41*	80	80
	(.73)	(.73)	(.83)	(.83)
R^2	.04	.04	.00	.00
Nat. output measure	1.50***	.38***	1.29**	.44**
	(.53)	(.12)	(.59)	(.21)
Output measure × cultural barrier	02	.14	00	.43
	(1.30)	(.30)	(1.29)	(.59)
R^2	.04	.04	.00	.00
Cultural barrier: pop-weighted distance	.47	.47	15	15
	(.39)	(.39)	(.18)	(.18)
R^2	.04	.04	.00	.00
Nat. output measure	1.47***	.40***	1.25**	.39**
	(.45)	(.11)	(.49)	(.18)
Output measure × cultural barrier	63	28***	81*	28*
	(.39)	(.08)	(.42)	(.14)

Table 10 Cultural similarities

Table 10 continued				
	(1)	(2)	(3)	(4)
R^2	.04	.04	.00	.00
Border-year fixed effects	Yes	Yes	Yes	Yes
No. observations	5408	5408	1352	1352
No. borders	270	270	270	270

Data on lights and population available from the NOAA and CIESIN, respectively, for 1990 (1992 for lights), 1995, 2000, 2005 and 2010. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account two-step estimation and infill asymptotics in parentheses. Each observation weighted by total population in the respective buffer piece. For each cross-border cultural similarity measure, the top line presents the results from regression (8), while the bottom two lines present results from regressions (9) or (10). Data on genetic distance from Spolaore and Wacziarg (2009) and data on shared languages, colonizers and population-weighted distance from CEPII (Mayer and Zignago (2011)). All border permeability variables standardized to have mean zero and unit variance. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account infill asymptotics in parentheses

tity) and macroeconomic policies. On the other hand, Glaser et al. (2004) and Gennaioli et al. (2013) argue that human capital explains cross-country (or cross-regional) income differences. Yet other researchers, for example Algan and Cahuc (2010), have hypothesized that differences in trust can explain differences in growth. It is therefore interesting to examine whether these factors help explain the discontinuities in growth rates at national borders that we have observed, although these results are quite tentative as there are no additional sources of identification, and the recovered correlations cannot be taken as causal.

The regressions that I run in this section will take the form:

$$\tilde{g}_{i,b,t} = \alpha_{b,t} + Z_{i,b}\Phi + \varepsilon_{i,b,t}$$
(13)

where $Z_{i,b}$ is a vector of covariates of country *i* (implicitly at border *b*, but these covariates are country-level and are not different border by border), Φ is a vector of coefficients on these covariates, and the remaining notation is the same as in Sect. 3.2^{7} This equation can be rewritten more intuitively as follows:

$$\Delta \tilde{g}_{i,b,t} = \Delta Z_{i,b} \Phi + \Delta \varepsilon_{i,b,t} \tag{14}$$

Hence, this is just a regression of the difference of growth rates at the border on the differences of possible national-level correlates of discontinuities between the two bordering countries. The covariates $Z_{i,b}$ can be thought of as a way of parametrizing the faster national growth rate indicator $u_{i,b,t}$ or the national growth rate $g_{i,t}$ from previous specifications.

Table 11, Panel 1 presents an analysis of the correlations of nighttime lights 5-year growth rates at the border with measures of institutions, public goods provision (paved roads), education and trust. Throughout the analysis, all right hand-side variables are normalized to have mean zero and unit variance, so the magnitudes of the coefficients are comparable. In the

²⁷ Results for growth in light density are somewhat weaker and statistically insignificant unless controls for population, area, climate and local public goods are included, but show the same pattern of coefficients as the results presented, and become numerically similar to results presented once the additional controls specified are included. This is most likely because my growth of light density at a border variable is uncalibrated to GDP, unlike my growth of lights per capita variable. Using growth of calibrated light density at a border restores significance to the rule of law coefficient estimates. Results for growth in light density are included in the Online Appendix I.

Table 11 Correlates of borde	er discontinuities							
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Specification	Rule of Law	Pav. Rd.	Educ	Trust	Ctrl Pav. Rd.	Ctrl Educ	Ctrl Trust	Ctrl All
Panel I: Dep. Var. is 5-Year gr	owth per capita at bor	der						
Rule of Law, WB	1.52 * * *				1.23 * * *	2.07***	2.22***	2.17^{***}
	(.47)				(.30)	(.42)	(.45)	(.30)
Other Control		1.32^{**}	.26	.66**	.63	74	.33	
		(.62)	(.46)	(.30)	(.44)	(.53)	(.23)	
R^2	.01	.01	00.	00.	.02	.03	.04	.05
Panel II: Dep. Var. is 18-Year	growth per capita at b	order						
Rule of Law, WB	1.52 * * *				1.19^{***}	2.04***	2.14***	2.01^{***}
	(.50)				(.30)	(.50)	(.62)	(.42)
Other control		1.38**	.35	.50**	.70	68	.18	
		(.65)	(.40)	(.22)	(.45)	(.51)	(.19)	
R^2	.08	.05	00.	.01	60.	.16	.17	.21
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1342	1336	974	808	1336	974	802	700
No. borders	266	264	186	159	264	186	156	130
Data on lights and population specification is the Eq. (3) lo determinants of growth descri border and taking into account	available from NOA, cal linear estimate of bed in Sect. 5. All cov t infill asymptotics in	A and CIESIN. Each the indicated con ariates except the parentheses. Each	ch panel of this cept of growth faster growth ra observation we	table presents of lights at the the indicator nor sighted by total	estimates of Eq. (13) for border corresponding malized to have mean (population in the 70-kr	or various covariate to the given count) and variance 1. Ro n neighborhood arc	s. The dependent va ry-border piece yea obust standard error, und the buffer piece	riable in each r. Data on all s clustered on

first column, I regress growth of nighttime lights at the border on border-piece by year fixed effects and on the quality of the rule of law as measured using the methodology of Kaufmann et al. (2009) in the World Governance Indicators. Recall that if there are no border discontinuities (growth in nighttime lights is the same on both sides of borders except for random noise), then the border-piece by year fixed effects should take out all the variation from the dependent variable, and there should be no correlation between nighttime lights at the border and the rule of law. However, I find that the rule of law has a coefficient of 1.52, which is statistically significant at 1 %. This means that a one standard deviation increase in the rule of law measure increases the average annualized growth rate of lights at that country's border by 1.52 percentage points. Columns 2 through 4 report similar univariate relationships of growth in nighttime lights at the border for several other potential determinants of growth. We see that the fraction of roads that are paved, and the fraction of survey respondents saying that "most people can be trusted" (Inglehart and Welzel 2008) are correlated with growth in nighttime lights at the border (although their coefficients are somewhat lower), but the average years of schooling measure of Barro and Lee (2010) is not. Columns 5 through 7 run horse races of the growth determinants in columns 2 through 4 against the rule of law variable. In all three regressions, the coefficient on the rule of law remains statistically significant, and of larger or comparable magnitude to column 1, while the coefficients on the other growth determinants fall relative to the univariate regressions and are never statistically significant. In column 8, we see that including all three growth determinants together as controls also does not weaken the rule of law coefficient. Panel 2 provides results for 18-year growth rates, which are essentially the same. Hence, of the potential determinants of growth considered, the rule of law appears to be robustly correlated with growth of nighttime lights at the border while the other determinants are not.

We further investigate the association between growth of nighttime lights at the border and the rule of law in Table 12, in which we experiment with alternative specifications and with more controls. Column 1 reproduces the baseline. Column 2 replicates the baseline regression without population weighting. We see that the coefficient on the rule of law variable declines by about a third of its magnitude, but remains statistically significant at 1 % and large. Column 3 replaces the rule of law variable with an indicator for the country having the larger value of the rule of law of the two countries at the given border. We see that having the larger value of the rule of law of the two bordering countries is associated with a 1 % higher average annual growth rate of lights at that border, although the statistical significance of those estimates is somewhat reduced. Column 4 removes outliers from the regression (all border pieces with absolute difference in growth rates per capita or national institutions greater than the 99th percentile of these variables in the baseline sample), which does not change the coefficient relative to column 1. Column 5 includes polynomials in population and area at the border, as well as climate and local public goods variables at the border, also with little change to the coefficient. Column 6 includes the spatial controls in column 5, as well as a bevy of additional control variables relating to contracting institutions, the size of government, the disease environment, regional controls and cultural controls, all of which only increases the coefficient on the rule of law variable. Lastly, column 7 replaces the rule of law variable with a combination of the expropriation risk measure from Acemoglu et al. (2001), where missing values of the measure have been imputed using the Fraser Institute's property rights index, and Fraser Institute's property rights index with missing values imputed from the expropriation risk measure. The coefficient on this alternative measure is somewhat lower than in column 1, but is still statistically significant and large.

It is interesting to ask how the association between the rule of law and border growth differences varies by continent. Table 13 presents estimates of Column 1 of Table 11 sepa-

Specification	(1)	(2) No weights	(3) Indicator only	(4) No out-liers	(5) Spatial controls	(6) All Ctrls	(7) Alt. Inst.
Panel I: Dep. Var. is 5-Year gro	wth per capita at t	border					
Rule of Law, WB	1.52^{***}	1.06^{***}	1.14^{**}	1.59^{***}	1.41^{***}	1.76^{***}	1.18^{**}
	(.47)	(.35)	(.58)	(.52)	(.35)	(.68)	(.51)
R^2	.01	00.	.00	.02	.05	.18	.01
Panel II: Dep. Var. is 18-year g	rowth per capita a	t border					
Rule of Law, WB	1.52^{***}	1.04^{***}	1.04^{*}	1.61^{***}	1.56^{***}	2.85***	1.17^{**}
	(.50)	(.36)	(.59)	(.52)	(.31)	(.68)	(.52)
R^2	.08	.02	.03	.12	.28	.71	.05
Border-Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. Observations	1342	1342	1352	1304	1342	588	1342
No. Borders	266	266	270	264	266	101	266
Data on lights and population : specification is the Eq. (3) loc determinants of growth descrit on border and taking into accou 6 contains all controls mention Freedom House's Political Riton	available from NO al linear estimate bed in Sect. 5. All unt infill asymptot wed so far as well a his Index Column	AA and CIESIN. Eacl of the indicated conce covariates except the 1 ics in parentheses. Eac as controls for genetic	h panel of this table pre- pt of growth of lights : faster growth rate indic: th observation weighted distance from the UK, t	sents estimates of Eq. at the border correspont ator normalized to hav by total population in he time to enforce a c multiv instead of the	(13) for various covariate anding to the given count e mean 0 and variance 1. the 70-km neighborhood ontract, the Malaria Ecolo and of law	s.The dependent va ry-border piece yes . Robust standard er I around the buffer f ogy index from Sac	riable in each r. Data on all rors clustered iece. Column hs (2002) and

Table 13 Correlates of b.	order disco	ntinuities									
Sample	(1) World	(2) No Asia	(3) No Africa	(4) No America	(5) No post-comm.	(6) Old borders	(7) Only Asia	(8) Only Africa	(9) Only America	(10) Only post- comm.	(11) New borders
Panel I: Dep. Var. is 5-Yea	ur growth pe	er capita at	border								
Rule of law, WB	1.64^{***}	1.13^{***}	1.75^{***}	1.70^{***}	1.65^{***}	1.64^{***}	1.92^{***}	1.24^{**}	.78	1.45*	1.32
	(.51)	(.43)	(.62)	(.52)	(.54)	(.52)	(.70)	(.52)	(1.17)	(.75)	(1.26)
R^2	.01	00.	.02	.02	.01	.01	.03	00.	00.	00.	.00
Panel II: Dep. Var. is 20-Y	ear growth	per capita	at border								
Rule of law, WB	1.63^{***}	1.29^{***}	1.72^{***}	1.64^{***}	1.64^{***}	1.65^{***}	1.82^{**}	1.32^{**}	1.41	1.34^{*}	.59
	(.54)	(.47)	(.67)	(.56)	(.56)	(.54)	(.76)	(.59)	(1.34)	(67.)	(1.09)
R^2	.08	.03	.13	60.	60.	60.	.16	.03	.05	.04	.00
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1342	1012	904	1144	1082	1190	330	438	198	260	152
No. borders	266	203	174	231	210	239	63	92	35	56	27
Data on lights and populat in each specification is the all determinants of growth Robust standard errors clu around the buffer piece	tion availab E Eq. (3) loo I described Istered on b	le from NO cal linear es in Sect. 5 oorder and t	AA and CIE timate of the All regions d aking into ac	SIN. Each pane indicated conc effned in Sect. count infill asy	l of this table pressept of growth of 1 4. All covariates e mptotics in parentl	ents estimates (ights at the bou except the faste heses. Each ob	of Eq. (13) e der correspo r growth rat servation we	stimated ove onding to the e indicator n eighted by to	r different regi given country ormalized to h tal population	ons. The depend border piece y ave mean 0 and in the 70-km n	lent variable ear. Data on I variance 1. sighborhood

rately for each continent, as well as for the world as a whole with continents dropped one by one. We see that the rule of law coefficient is robust to dropping any continent, and that it is quantitatively large (around unity or greater) when the regression is performed on each continent alone except the Americas in the 5-year panel. We do see, however, that the rule of law coefficient loses significance if we restrict the estimation sample to be composed of borders that were drawn only recently (the borders between the post-Soviet states, and within the former Yugoslavia and Czechoslovakia), and shrinks substantially if we look at new borders in the 18-year panel.

The regressions in Tables 11, 12, and 13 differ from the specifications in Sects. 4 and 5. In the previous sections, I have investigated how border discontinuities change in magnitude depending on a characteristic of the border, such as its shape, its date of creation, how much trade goes across it, and whether the people on the opposite sides of it have the same or different culture. Here, however, I consider characteristics not of the border itself, but of the bordering countries, such as their institutional or educational endowments. While it is possible to convert characteristics of bordering countries into characteristics of the border (for example, by constructing a measure of difference between institutions at a border, and running a regression with it as a border barrier measure, in the manner of Sect. 5), this is not the most efficient use of the data, because it ignores information about which country has the greater and which country has the lesser value of the characteristic in question. For example, if I use the difference between institutions at the border as a barrier measure, I lose any data on which country has better institutions (and might be expected to have more light at the border) and which country has worse institutions (and may be expected to have less light at the border). However, running these specifications is useful to assess to what extent cross-country differences in various variables explain the magnitude (rather than the direction) of border discontinuities.

In Table 14 I present estimates from specifications (8), (9) and (10) in which the crossborder flow measure T_b is defined to be the absolute value of the difference in institutions between the bordering countries for the first panel, the absolute value of the difference in average years of schooling for the second panel, and so on. There are virtually no effects of the cross-country difference in institutions, roads or trust on the size of the growth discontinuity at the border between these countries. Larger educational differences between bordering countries are actually associated with smaller growth discontinuities at borders. In the last panel of the table, I use the absolute value of the differences in national 18-year growth rates as the measure T_b . While this absolute value is strongly correlated with the absolute value of the difference of growth at the border, it is not correlated with the growth rate of the fastergrowing side of the border. Therefore, using absolute values of differences of covariates as indicators of border barriers leaves out a lot of the information that these covariates contain.

We learn from Table 14 that institutions have explanatory power for the direction of the growth discontinuity at a border (the sides of borders with better institutions have faster growth at borders), but the magnitude of the institutional gap at a border does not predict the magnitude of the gap in growth rates at that border. In particular, variation in institutions still leaves a large fraction of variation in growth discontinuities at borders unexplained. One rationalization for this pattern could be that the magnitudes of institutional differences are measured poorly, but the signs of institutional differences are measured relatively well (the latter being consistent with Row 2 of Table 12). This could happen, for example, if measures of freedom or political risk for each of two neighboring countries are constructed by people who know both of the neighboring countries much better than they know the rest of the world, so they get the ranking right between the neighboring countries, but not necessarily

	(1)	(2)	(3)	(4)
Dep. Var.	5-Year growth p/c	5-Year growth p/c	18-Year growth p/c	18-Year growth p/c
Ind. Var.	Dummy	Elast.	Dummy	Elast.
Absolute difference:	25	25	.10	.10
institutions	(.44)	(.44)	(.31)	(.31)
R^2	.04	.04	.00	.00
Nat. output measure	1.48***	.38***	1.28**	.46**
	(.51)	(.12)	(.56)	(.19)
Output measure × Absolute Difference	06	03	.02	05
	(.55)	(.12)	(.59)	(.21)
R^2	.04	.04	.00	.00
Absolute difference: paved roads	03	03	.16	.16
	(.46)	(.46)	(.24)	(.24)
R^2	.04	.04	.00	.00
Nat. output measure	1.48***	.38***	1.28**	.45**
	(.51)	(.12)	(.56)	(.20)
Output measure × Absolute Difference	08	02	.08	01
	(.58)	(.13)	(.65)	(.26)
R^2	.04	.04	.00	.00
Absolute difference: education	.00	.00	36*	36*
	(.36)	(.36)	(.20)	(.20)
R^2	.04	.04	.00	.00
Nat. output measure	1.48***	.36***	1.27**	.44**
	(.44)	(.11)	(.50)	(.19)
Output measure × Absolute Difference	92**	18*	80**	19
	(.41)	(.10)	(.36)	(.14)
R^2	.04	.04	.00	.00
Absolute difference: trust	1.04	1.04	13	13
	(.68)	(.68)	(.20)	(.20)
R^2	.06	.06	.00	.00
Nat. output measure	1.49***	.38***	1.27**	.44**
-	(.49)	(.11)	(.55)	(.20)
Output measure × Absolute Difference	43	02	39	10
	(.32)	(.09)	(.42)	(.12)
<i>R</i> ²	.06	.06	.00	.00

 Table 14
 Absolute differences in covariates

Table 14 continued

	(1)	(2)	(3)	(4)
Absolute difference: growth	1.97***	1.97***	.64**	.64**
	(.42)	(.42)	(.27)	(.27)
R^2	.10	.10	.02	.02
Nat. output measure	1.48***	.38***	1.28**	.47**
	(.52)	(.12)	(.56)	(.21)
Output measure × Absolute Difference	.11	.00	14	16
	(.44)	(.11)	(.46)	(.19)
R^2	.10	.10	.02	.02
Border-year fixed effects	Yes	Yes	Yes	Yes
No. observations	5408	5408	1352	1352
No. borders	270	270	270	270

Data on lights and population available from the NOAA and CIESIN, respectively, for 1990 (1992 for lights), 1995, 2000, 2005 and 2010. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account two-step estimation and infill asymptotics in parentheses. Each observation weighted by total population in the respective buffer piece. Each absolute difference is the absolute value of the difference of the national-level covariate indicated across the given border. The top line presents the results from regression (8), while the bottom two lines present results from regressions (9) or (10). All absolute differences standardized to have mean zero and unit variance. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account infill asymptotics in parentheses

between those countries and other, non-neighboring countries. Such a pattern of data collection is plausible because many organizations that measure institutions make use of regional experts.

6.2 Local Variation in Public Goods

While most of the determinants of growth that I use in Sect. 6.1 are observable only at the country level, extensive geographical information exists for the provision of certain public goods, namely roads, railroads and utility lines. Therefore, I can directly check whether differences in the levels of public goods provision around a border may be explaining discontinuities in economic growth at that border by looking at whether growth discontinuities coincide with discontinuities in public goods, and whether faster-growing sides of borders have sharply higher levels of public goods than slower-growing sides of borders. An example of public goods driving income differentials is found by Dell (2010), who documents that road density falls discontinuously as one crosses a border into a Peruvian region in which forced labor was practiced during colonial times, and provides anecdotal evidence that road quality in that region is critical for access to markets.

I obtain spatial data on roads, railroads and utility lines (power and telephone lines) from the Digital Chart of the World (DCW). In results not reported, I confirm that the national aggregates of roads and railroads match up well with national-level statistics presented by the World Bank, and that there is a strong positive association between the log total public goods in a fixed neighborhood of a border and the amount of lights per capita in that neighborhood.²⁸ The DCW attempts to convey information on road quality by classifying roads as "primary or secondary roads," "divided highways" and "paths or trails." Presumably, trails are of lower quality than are the other two types of roads, so the fraction of roads that are not trails should be a reasonable measure of road quality.

Table 16 presents estimates of growth discontinuities at borders with above- and belowmedian differentials in public goods density (or quality), in the same way as Tables 4 and 5 presented these estimates for geographic border differentials. It is apparent that growth discontinuities are present at borders regardless of the size of the public goods differential across the border, except for borders without large differentials in utility lines in the case that national growth rates are the independent variable. Therefore, Table 16 provides some weak evidence that growth discontinuities at borders might be affected by differentials in public goods provision. Hence, in Row 6 of Table 3 I include local public goods variables for the log lengths of roads, railroads and utility lines (and for the quality of the roads) in each border piece neighborhood as controls in my baseline set of regressions. Including local public goods variables does not change the discontinuities in economic growth.

Another way to see that variation in public goods does not account for growth discontinuities is to ask whether sides of borders corresponding to faster-growing countries have higher public goods densities. In Table 15, I obtain estimates of discontinuities in the amounts of these public goods variables per capita across national borders, which I construct using the two-step local linear procedure described in Sect. 4. I normalize the dependent variable to have mean zero and unit variance in all specifications. The first two rows present baseline estimates for border pieces with positive amounts of infrastructure in a 70-km neighborhood. We see that road density is not systematically higher on the sides of borders corresponding to faster-growing countries (the estimates of the difference are positive, as might be expected if higher road density is associated with higher economic growth or development, but they are quantitatively small and statistically insignificant). However, as we see in column 2, the fraction of roads that are not trails does not vary discontinuously across borders. Column 3 of Table 15 presents the results for railroads. The estimates are larger, but not significant. Finally, column 4 presents the same analysis with the dependent variable being log total length of utility lines (most frequently these are power lines, but also included are telephone lines and pipelines), which also do not show a tendency to be more dense on faster-growing sides of borders and, if anything, may show the opposite tendency.

In rows 3 and 4, I add quartic polynomials in the population and the area of the neighborhood around the border piece, as well as local linear estimates of geographic variables (described in Sect. 3) at each side of the border to the regressions above. Langaas (1995) suggests that the Digital Chart of the World may be biased because it reports infrastructure primarily for the purpose of helping pilots identify their location, and underreports infrastructure in built-up areas. To the extent that slower-growing sides of borders have fewer built-up areas than faster-growing sides of borders do, such a bias may lead to attenuate discontinuities at borders. While we cannot remove this bias directly, a weak test of the importance of the bias is to add the geographic variables as well as flexible population and area controls. With them, I should be able to proxy for built-up area to some extent and at least move in the direction of the true discontinuity by alleviating this bias. We see that adding these controls,

 $^{^{28}}$ In particular, there is a 0.85 correlation between total road length in the DCW and total road length as reported by the World Bank. There is a 0.55 correlation between total telephone line length in the DCW and total telephone line length as reported by the World Bank.

Border-year fixed effects

No. observations

No. borders

 R^2 of last row

Table 15 Benavior of local public goods	across borders			
	(1)	(2)	(3)	(4)
Dep. Var.	Log	Fraction	Log	Log
	Roads	Roads	Railroads	Utilities
5-Year growth rate	.03	00	.07	05
	(.02)	(.00)	(.04)	(.08)
20-Year growth rate	.02	00	.11	03
	(.04)	(.01)	(.07)	(.10)
5-Year growth rate, Controls	.00	.00	00	.00
	(.02)	(.00)	(.02)	(.02)
20-Year growth rate, Controls	00	.00	.06	.01
	(.03)	(.00)	(.04)	(.04)
5-Year growth rate, dummy, controls	01	.00*	.00	.00
	(.00)	(.00)	(.01)	(.01)
20-Year growth rate, dummy, controls	.00	.01**	.00	.00
	(.01)	(.00)	(.01)	(.01)

Yes

1352

270

.00

Yes

1352

270

.03

 Table 15 Behavior of local public goods across borders

Data on lights and population available from NOAA and CIESIN. Data roads, railroads, utilities and fractions of roads that are not trails from the Digital Chart of the World. Each column presents estimates of Eq. (4) or (5) with local public goods variables replacing economic growth. The dependent variable for the first four rows is the Eq. (3) local linear estimate of the log amount or fraction of specified public good provided that it is not zero in the relevant buffer piece. The dependent variable for the last two rows is an indicator that the amount of the specified public good is not zero. All public goods variables rescaled to have mean 0 and variance 1. Observation unit is a country-border piece-year. Robust standard errors clustered on border and taking into account infill asymptotics in parentheses. Each observation weighted by total population in the 70-km neighborhood around the buffer piece. The independent variable, whose coefficient is given in each cell, is the indicator that a given side of the border corresponds to the country with the higher corresponding national growth rate (5-year or 18-year) of GDP per capita from WDI

Yes

1352

270

.00

if anything, brings the estimates of discontinuities in local public goods closer to zero. For many border pieces, there is no infrastructure in 70-km neighborhoods of borders, and I have previously left them out of our analysis. In rows 5 and 6, I change my dependent variable to a dummy for the presence of particular types of infrastructure at the border to see whether there may be discontinuities in infrastructure at the extensive margin. However, these discontinuity estimates also turn out to be small and (except for the fraction of roads that are not trails) insignificant.

It is instructive to look at maps of road density for the regions in which we observed significant discontinuities in the growth rate of nighttime lights per capita. In contrast, these regions appear to have continuous road density across borders. Figure 2 shows that in Eastern Europe, road networks rarefy in the sparsely populated Carpathian mountain region that straddles Ukraine and Romania, but have the same density on either side of the Ukraine-Poland or the Romania-Moldova borders. Moreover, utility lines appear to be denser in northern Ukraine than in nearby Poland (or eastern Czech Republic), despite the fact that Poland has had a much stronger growth experience than did Ukraine during our sample

Yes

1352

270

.00

TADIE 10 OTOWII UISCONII	numes by bord	er type							
Border Type	(1) All jump	(2) Roads jump	(3) No Roads jump	(4) Frac Paved jump	(5) No Frac Paved jump	(6) RailRds jump	(7) No RailRds jump	(8) Utilities jump	(9) No Utilities jump
Dummy, 5-Year growth	1.48***	1.78^{***}	1.18^{**}	1.42	1.55***	.18	2.80***	1.82^{**}	1.13^{**}
Lights per capita	(.51)	(89)	(.55)	(.87)	(.53)	(.53)	(.56)	(.88)	(.53)
5-Year growth	.38***	.40***	.35**	.34	.40***	.13	.80***	.52***	.24
Lights per capita	(.11)	(.14)	(.17)	(.22)	(.13)	(11)	(.20)	(.16)	(.15)
Dummy, 18-Year Growth	1.28^{**}	1.54^{**}	1.01	1.54*	1.00*	.22	2.31^{***}	1.79^{**}	.70
Lights per capita	(.57)	(.67)	(.73)	(.91)	(.56)	(.50)	(.74)	(.81)	(.53)
18-Year growth	.45**	.52**	.36	.48	.41**	03	.92***	.71**	.15
Lights per capita	(.20)	(.23)	(.30)	(.37)	(.17)	(.20)	(.29)	(.29)	(.19)
Border-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	1352	902	450	648	704	640	712	576	776
No. borders	270	230	150	179	166	178	190	168	201
\mathbb{R}^2 of first row	.04	.05	.02	.05	.03	00.	.18	.12	00.
Data on lights and populati fractions of roads that are 1 exceeds the population-wei eurors clustered on border buffer piece. The dependen country-border piece year. higher corresponding nation	on available from not trails from ghted median (and taking into t variable in eau The independe. nal growth rate	om the NOAA is the Digital Change of these differences of these differences account two-second two-second translable, who is the theory of the specification of the theory of the the theory of the theory of theory of the theory of the theory	and CIESIN, res art of the World. res. Each colurn step estimation a 1 is the Eq. (3) to ose coefficient is year) of GDP per	pectively, for 19 A "jump" is de in presents estin nd infill asympt cal linear estimic s given in each o	90 (1992 for light fined to take plac, nates of Eq. (4) or totics in parenthes are of the indicated cell, is the indicated DI, or the correspo	s), 1995, 2000, e whenever the (5). Observation ces. Each observ d concept of gro or that a given s onding national	2005 and 2010. De difference in the c n unit is a country- vation weighted by with of lights at the ide of the border to growth rate itself	ata on roads, rail limatic variable border piece-yes / total population e border correspo corresponds to th	roads, utilities and for a border piece r. Robust standard 1 in the respective anding to the given e country with the

period. Hence, it is plausible that local public goods networks are continuous across borders more generally, and hence, that growth discontinuities at borders are not accounted for by variation in them.

7 Conclusion

This paper uses satellite data on nighttime lights to find large and statistically significant discontinuities in 5-year and 18-year growth rates of economic activity at borders. To my knowledge, discontinuities in economic growth have not been explored for the world as a whole in the economics literature. These discontinuities are not driven by any particular region, nor can they be easily explained by obvious differences at borders in terrain, climate or public goods provision. I consider a framework in which growth discontinuities in economic activity are consistent either with borders acting as barriers to determinants of growth, such as institutions, culture or human capital, or with borders acting as barriers to the diffusion of economic activity even in a situation in which the determinants of growth on both sides of the border are the same. Using data on trade, migration and remittances between countries, I provide evidence that barriers to trade play a small role in generating growth discontinuities because the magnitudes of growth discontinuities do not seem to respond to differences in trade flows across the borders, but growth discontinuities often arise at borders between culturally dissimilar countries. Finally, I present a correlational analysis of potential determinants of border discontinuities, and find that differences in the quality of the rule of law across borders has substantial power in explaining the directions (though not the magnitudes) of discontinuities in economic growth rates at borders, as compared to variables measuring culture or public goods provision.

Acknowledgements I would like to thank Daron Acemoglu, Jerry Hausman, James Snyder and Amy Finkelstein for guidance and advice. I would like to thank Melissa Dell, Esther Duflo, Susan Elmes, Chris Elvidge, Jens Hainmueller, Horacio Larreguy, Anna Mikusheva, Serena Ng, Elias Papaioannou, Jack Porter, Bernard Salanie, Whitney Newey, and participants at the Political Economy Breakfast at MIT and the Econometrics Lunch at Columbia University for very insightful comments. I am very grateful to Oded Galor and four anonymous referees for insightful and useful comments. I would like to thank Daniel Sheehan and the MIT GIS lab for assistance with geographic software. I am very grateful to John Grigsby for truly superb and dedicated research assistance. I am very grateful to the Paul and Daisy Soros Fellowship for New Americans for intellectual stimulation, and to the National Science Foundation Graduate Research Fellowship Program and the Institute of Humane Studies for funding. The paper does not necessarily represent the views of the above individuals and organizations, of the Federal Reserve Bank of New York, or of the Federal Reserve System. All remaining errors are my own.

References

- Acemoglu, D. (2005). Politics and economics in weak and strong states. Journal of Monetary Economics, 52, 1199–1226.
- Acemoglu, D., & Dell, M. (2010). Productivity differences between and within countries. American Economic Journal: Macroeconomics, 2(1), 169–188.
- Acemoglu, D., Gallego, F. A., & Robinson, J. A. (2014). Institutions, human capital, and development. Annual Review of Economics, 6(1), 875–912.

Acemoglu, D., & Johnson, S. (2005). Unbundling institutions. Journal of Political Economy, 113(5), 949-995.

Acemoglu, D., Johnson, S., & Robinson, J. A. (2000). The colonial origins of comparative development: An empirical investigation. NBER working papers: 7771.

Acemoglu, D., Johnson, S., & Robinson, J. A. (2001). The colonial origins of comparative development: An empirical investigation. *American Economic Review*, 91(5), 1369–1401.

- Acemoglu, D., Johnson, S., & Robinson, J. A. (2002a). Reversal of fortune: Geography and institutions in the making of the modern world income. *Quarterly Journal of Economics*, 117(4), 1231–1294.
- Acemoglu, D., Johnson, S., Robinson, J. A., & Thaicharoen, Y. (2003). Institutional causes macroeconomic symptoms: volatility, crises and growth. *Journal of Monetary Economics*, 50, 49–123.
- Acemoglu, D., Johnson, S., Robinson, J. A., & Yared, P. (2008). Income and democracy. American Economic Review, 98(3), 808–842.
- Alesina, A., Easterly, W., & Matuszeski, J. (2011). Artificial states. Journal of the European Economic Association, 9, 246–277.
- Alesina, A., Michalopoulos, S., & Papaioannou, E. (2016). Ethnic inequality. Journal of Political Economy, 124(2), 428–488.
- Algan, Y., & Cahuc, P. (2010). Inherited trust and growth. American Economic Review, 100(5), 2060–2092.
- Banerjee, A., & Iyer, L. (2005). History, institutions, and economic performance: The legacy of colonial land tenure systems in India. American Economic Review, 95(4), 1190–1213.
- Barro, R. J. & Lee J.-W. (2010). A new data set of educational attainment in the world, 1950–2010. NBER working papers: 15902.
- Baugh K. E., Elvidge C. D., Ghosh T. & Ziskin D. (2009). Development of a 2009 stable lights product using DMSP-OLS data. In Proceedings of the 30th Asia–Pacific advanced network meeting.
- Bester, A. C., Conley, T. G., Hansen, C. B., & Vogelsang, T. J. Fixed-b asymptotics for spatially dependent robust nonparametric covariance matrix estimators.
- Center for International Earth Science Information Network (CIESIN), Columbia University; and Centro Internacional de Agricultura Tropical (CIAT). 2005. Gridded population of the world, version 3 (GPWv3). Palisades, NY: Socioeconomic Data and Applications Center (SEDAC), Columbia University. Retrieved from http://sedac.ciesin.columbia.edu/gpw.
- Center for International Earth Science Information Network (CIESIN). (2013). Anthropogenic biomes of the world, version 2. Palisades, NY: Socioeconomic Data and Applications Center (SEDAC), Columbia University. Retrieved from http://sedac.ciesin.columbia.edu/data/set/anthromes-anthropogenic-biomesworld-v2-2000.
- Chen, Xi, and Nordhaus, William D. (2010). The value of luminosity data as a proxy for economic statistics. NBER working papers: 16317.
- Conley, T. G. (1999). GMM estimation with cross sectional dependence. Journal of Econometrics, 92(1), 1–45.
- Cressie, N. G. (1993). Statistics for spatial data. New York: Wiley.
- Cust, J. & Harding, T. (2014). Institutions and the location of oil exploration. Manuscript.
- de Soto, H. (2000). The Mystery of Capital. New York: Basic Books.
- Dell, M. (2010). The persistent effects of Peru's mining mita. Econometrica, 78(6), 1863–1903.
- Doll, C. N.H. (2008). CIESIN thematic guide to night-time light remote sensing and its applications. Manuscript.
- Doll, C. N. H., Muller, J. P., & Morley, J. G. (2006). Mapping regional economic activity from night-time light satellite imagery. *Ecological Economics*, 57, 75–92.
- Elvidge, C. (2003). Nighttime lights change detection. PowerPoint Presentation.
- Elvidge, C. D., Baugh, K. E., Kihn, E. A., Kroehl, H. W., & Davis, E. R. (1997). Mapping city lights with nighttime data from the DMSP operational linescan system. *Photogrammetric Engineering & Remote Sensing*, 63(6), 727–734.
- Elvidge, C. D., Baugh, K. E., Dietz, J. B., Bland, T., Sutton, P. C., & Kroehl, H. W. (1999). Radiance calibration of DMSP-OLS low-light imaging data of human settlements. *Remote Sensing of Environment*, 68(1), 77–88.
- Elvidge, C. D., Baugh, K. E., Sharolyn, J., Anderson, P., Sutton, C., & Ghosh, T. (2012). The Night Light Development Index (NLDI): A spatially explicit measure of human development from satellite data. *Social Geography*, 7, 23–35.
- Gennaioli, N., & Rainer, I. (2007). The modern impact of precolonial centralization in Africa. Journal of Economic Growth, 12(3), 185–234.
- Ghosh, T., Powell, R. L., Elvidge, C. D., Baugh, K. E., Sutton, P. C., & Anderson, S. (2010). Shedding light on the global distribution of economic activity. *The Open Geography Journal*, 3, 148–161.
- Glaeser, E. L., La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2004). Do institutions cause growth? *Journal of Economic Growth*, 9(3), 271–303.
- Gorodnichenko, Y., & Roland, G. (2016). Culture, institutions and the wealth of nations. *Review of Economics and Statistics* (Forthcoming).
- Hahn, J., Todd, P., & Van der Klaauw, W. (2001). Identification and estimation of treatment effects with a regression-discontinuity design. *Econometrica*, 69(1), 201–209.

- Henderson, V., Storeygard, A., & Weil, D. N. (2011). A bright idea for measuring economic growth. American Economic Review, 101(3), 194–199.
- Heston, A., Summers, R. & Aten, B. (2011). Penn world table Version 7.0, Center for international comparisons of production, income and prices at the University of Pennsylvania, May 2011.
- Hijmans, R. J., Cameron, S. E., Parra, J. L., Jones, P. G., & Jarvis, A. (2005). Very high resolution interpolated climate surfaces for global land areas. *International Journal of Climatology*, 25, 1965–1978.
- Huillery, E. (2009). History matters: The Long-term impact of colonial public investments in French West Africa. American Economic Journal: Applied Economics, 1(2), 176–215.
- Imbens, G. W., & Lemieux, T. (2008). Regression discontinuity designs: A guide to practice. Journal of Econometrics, 142(2), 615–635.
- Inglehart, R., & Welzel, C. (2008). The WVS cultural map of the world. Retrieved 9 March, 2013 from http:// www.worldvaluessurvey.org/wvs/articles/folder_published/article_base_54.
- International Monetary Fund. (2005). *Direction of trade statistics*. Washington, D.C.: Database and Browser CD-ROM.
- Jerven, M. (2013). Poor numbers: How we are misled by African development statistics and what to do about it. Cornell University Press.
- Johnson, S., Larson, W., Papageorgiou, C., & Subramanian, A. (2009). Is newer better? Penn world table revisions and their impact on growth estimates. NBER working papers: 15455
- Kaufmann, D., Kraay, A., & Mastruzzi, M. (2009). Governance matters VIII: Aggregate and individual governance indicators, 1996–2008. World Bank Policy Research Working Paper 4978.
- Larreguy, H. (2011). The lasting impact of colonial policies in Nigeria: Evidence from a natural experiment. Manuscript.
- Lee, D. S., & Card, D. (2008). Regression discontinuity inference with specification error. *Journal of Econo*metrics, 142(2), 655–674.
- Lee, D. S., & Lemieux, T. (2010). Regression discontinuity designs in economics. Journal of Economic Literature, 48(2), 281–355.
- Matheron, G. (1963). Principles of geostatistics. Economic Geology, 58, 1246-1266.
- Mayer, T. & Zignago, S. (2011). Notes on CEPII's distances measures (GeoDist), CEPII working paper 2011-25. Retrieved 1 March, 2014 from http://www.cepii.fr/anglaisgraph/bdd/distances.htm#sthash. MwBjhazf.dpuf
- McCallum, J. T. (1995). National borders matter: Canada-U.S. regional trade patterns. American Economic Review, 85(3), 615–623.
- Medrano, J. D. (2008). World map of interpersonal trust. Banco de Datos ASEP / JDS. Retrieved 1 March, 2014 from http://www.jdsurvey.net/jds/jdsurveyMaps.jsp?Idioma=I&SeccionTexto=0404&NOID=104.
- Michalopoulos, S., & Papaioannou, E. (2014). National institutions and sub-national development in Africa. Quarterly Journal of Economics, 129(1), 151–213.
- Newman, C. M. (1984). Asymptotic independence and limit theorems for positively and negatively dependent random variables. Lecture Notes-Monograph Series, Inequalities in statistics and probability (Vol. 5,, pp. 127–140)
- Nunn, N., & Puga, D. (2012). Ruggedness: The blessing of bad geography in Africa. Review of Economics and Statistics, 94(1), 20–36.
- Pakes, A. & Griliches, Z. (1980). Patents and R and D at the firm level: A first look. NBER working papers: 0561.
- Pinkovskiy, M., & Sala-i-Martin, X. (2016). Lights, camera, income! illuminating the national accountshousehold surveys debate. *Quarterly Journal of Economics*, 131(2), 579–631.
- Porter, J. (2003). Estimation in the regression discontinuity model. Manuscript.
- Robinson, P. M. (2011). Asymptotic theory for nonparametric regression with spatial data. *Journal of Econo*metrics, 165(1), 5–19.
- Rodrik, D., Subramanian, A., & Trebbi, F. (2004). Institutions rule: The primacy of institutions over geography and integration in economic development. *Journal of economic growth*, 9(2), 131–165.
- Sachs, J. D. (2002). Institutions don't rule: Direct effects of geography on per capita income. NBER working paper #9490.
- Schabenberger, O., & Gotway, C. (2005). Statistical methods for spatial data analysis. Boca Raton: Chapman and Hall/CRC.
- Spolaore, E., & Wacziarg, R. (2009). The diffusion of development. *Quarterly Journal of Economics*, 124(2), 469–592.
- Stein, M. L. (1987). A modification of minimum norm quadratic estimation of a generalized covariance function for use with large data sets. *Journal of the American Statistical Association*, 82(399), 765–772.
- Stein, M. L. (1999). Interpolation of spatial data : Some theory for kriging. New York: Springer.

Sutton, P. C., Elvidge, C. D., & Ghosh, T. (2007). Estimation of gross domestic product at sub-national scales using nighttime satellite imagery. *International Journal of Ecological Economics & Statistics*, 8, 5–21.

- Tabellini, G. (2010). Culture and institutions: Economic development in the regions of Europe. *Journal of the European Economic Association*, 8(4), 677–716.
- USGS. (2000). Shuttle radar topography mission. Global Land Cover Facility: University of Maryland, College Park, Maryland.
- U.S. National Imagery and Mapping Agency (NIMA). (2003). VMAP0DATA National Stock Number (NSN): 7644014361830, 7644014361831, 7644014361833. Fairfax., VA.
- Weidmann, N. B., Kuse, D., & Gleditsch, K. S. (2010). The geography of the international system: The CShapes dataset. *International Interactions*, 36(1), 86–106.
- World Bank. (2011). Migration and remittances factbook 2011. Washington, DC
- World Bank. (2011). World development indicators. Washington, DC.
- World Bank. (2011). World governance indicators. Washington, DC.